COLLECTIVE AGREEMENT

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Employer)

OF THE FIRST PART

-and-

UNIFOR, LOCAL 302 (hereinafter called the Union)

OF THE SECOND PART

ARTICLE 1	GENERAL PURPOSE		1
	1.1 Full-Time Employees Only	1	
	1.2 Part-Time Employees Only	1	
ARTICLE 2	CLIENTS OF THE HOME		2
ARTICLE 3	CONFLICTING AGREEMENTS AND DISCR	IMINATION	2
/IIIIII	3.3 Discrimination	2	
	3.4 Workplace Harassment	3	
	3.5 Violence Against Women	3	
	3.6 Staff Planning Consultation Committee	4	-
	3.7 Technological Change	5	
ARTICLE 4	COLLECTION OF UNION DUES OR UNION	SECURITY	5
ARTICLE 5	NO STRIKES OR LOCKOUTS		6
ARTICLE 6	RESERVATION OF EMPLOYER'S FUNCTIONS		6
ARTICLE 7	REPRESENTATION		7
	7.5 Full-Time Employees Only	8	
	7.7 Part-Time Employees Only	9	
	7.8 Health and Safety	9	
ARTICLE 8	GRIEVANCE PROCEDURE		11
3-11	8.1 Complaint Resolution Step	11	
	8.2 Grievance Procedure – Step 1	12	
	Grievance Procedure – Step 2	12	
	8.3 Referral to Arbitration – Step 3	12	
ARTICLE 9	ARBITRATION		13
ARTICLE 10	DISCHARGE CASES		15
ARTICLE 11	POLICY GRIEVANCES		15
	11.2 Policy Grievance defined as	16	
ARTICLE 12	SENIORITY		16
	12.3 Full-Time Employees Only	16	
	12.6 Part-Time Employees Only	17	
	12.9 Lay-off and Recall	18	

ARTICLE 13	JOB CLASSIFICATION AND RATES OF PAY		21
ARTICLE 14	HOURS OF WORK		21
	14.8 Full-Time Employees Only	23	
	14.10 Part-Time Employees Only	23	
ARTICLE 15	PAID HOLIDAYS		25
,	15.2 Full-Time Employees Only	25	
	15.3 Part-Time Employees Only	26	
ARTICLE 16	SICK LEAVE		28
	16.3 Full-Time Employees Only	29	
ARTICLE 17	VACATIONS		32
ANTIOLL II	17.0 Full-Time and Part-Time Employees	32	32
	17.1 Full-Time Employees Only	33	- ii-
<u> </u>	17.2 Part-Time Employees Only	35	
ARTICLE 18	COMPASSIONATE LEAVE AND LEAVE OF A 18.1 Bereavement Leave	ABSENCE 36	36
		37	
	18.2 Pregnancy and Parental Leave	[0.58074]	
	18.9 Full-Time Employees Only 18.10 Part-Time Employees Only	40	
	18.11 Jury Duty	40	
	18.12 Union Leave	41	
	18.13 Union Education Leave	41	
	18.14 Education Leave	42	
	18.15 General Leave	42	
ARTICLE 19	HOSDITAL MEDICAL DENSION INSUDANG	CE DROCRAM	42
ANTIGLE 13	HOSPITAL, MEDICAL, PENSION, INSURANCE 19.1 Full-Time Employees Only	42	42
	19.2 Part-Time Employees Only	45	
ARTICLE 20	REPORTING PROCEDURES		45
ARTICLE 21	POSTING OF STAFF VACANCY		46
ARTICLE 22	FILLING OF TEMPORARY FULL-TIME POSITIONS – PART-TIME EMPLOYEES ONLY		48
	22.1 Filling of Temporary Regular Part-Time Positions – Casual Part-Time Employees Only	49	
ARTICLE 23	UNIFORMS		50

ARTICLE 24	MEDICAL EXAMINATION		50
ANTIOLL 24	24.1 (a) and (b) Costs	50	50
	24.1 (a) and (b) costs	50	
ARTICLE 25	TERMINATION OF EMPLOYMENT		51
AITHOLL 20	TERMINATION OF EMPLOTMENT		- 31
ARTICLE 26	TRAINING AND EDUCATIONAL COURSE	S	51
ARTICLE 27	LETTERS OF REPRIMAND		51
ARTICLE 28	ADVICE TO THE EMPLOYER		51
ARTICLE 29	PERSONNEL FILES	-	51
ARTICLE 30	RETROACTIVE PAY		52
ARTICLE 31	MILEAGE ALLOWANCE		52
			- 111
ARTICLE 32	RPN ALLOWANCE		52
ARTICLE 33	TERM OF AGREEMENT	PARTIES ESSUES	52
SCHEDULE "A"	FULL-TIME AND PART-TIME		54
	SERVICE WORKERS WAGES		
SCHEDULE "B"	LONG TERM DISABILITY DETAILS		57
	LETTERS OF UNDERSTANDING		58
	Format of the Collective Agreement	59	
	Call-In and Overtime	60	
	Temporary Full-Time Hours	62	
	Emergency Call-in List	63	
	Scheduling Times and Rotation	64	
	Education Leave	66	
	Regular Part-Time and Casual Part-Time Resident Abuse	67	
	Women's Advocate	69	
	W 10 10 10 10 10 10 10 10 10 10 10 10 10	70	
	Outbreaks in the Workplace	71	
	Diversity and Inclusion Uniforms	72	-
		73	
	Summer Students – PSW/HCA	74	

COLLECTIVE AGREEMENT

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (hereinafter called the Employer)

OF THE FIRST PART

-and-

UNIFOR, LOCAL 302 (hereinafter called the Union)

OF THE SECOND PART

NOW THEREFORE THIS AGREEMENT WITNESSETH:

ARTICLE 1 - GENERAL PURPOSE

1.01 The definition of the Bargaining Units covered by this Collective Agreement are as more particularly described in Article 1.1 for Full – Time employees and Article 1.2 for Part – Time employees.

Full-Time Employees Only

1.1 The Employer recognizes the Union as the exclusive bargaining agent for all the Employer's employees who are employed at Dearness Home save and except: Office and Clerical Employees, Management Administrative Assistants, Supervisors, persons above the rank of Supervisor, Payroll Staff, summer students employed during the summer months and casual help working not more than 24 hours per week.

Part-Time Employees Only

1.2 The Employer recognizes the Union as exclusive bargaining agent for all the Employer's employees employed at Dearness Home save and except: Office and Clerical Employees, Management Administrative Assistants, Supervisors, persons above the rank of Supervisor and Payroll Staff.

ARTICLE 2 - CLIENTS OF THE HOME

2.0 This Agreement shall not apply to residents and clients of the Home who perform such services as therapy, provided, however, that this provision shall not interfere with the working conditions or the jobs of the employees covered by this Agreement, and the use of residents and clients as outlined above shall not be used to reduce the number of staff.

ARTICLE 3 - CONFLICTING AGREEMENTS AND DISCRIMINATION

- 3.0 (a) The Employer undertakes that it will not enter into any other Agreement with employees as herein defined, either individually or collectively, which will conflict with any of the provisions of the Agreement.
 - (b) Each of the Parties hereto agree that there will be no discrimination, interference, restraint or coercion exercised or practised upon any employee because of membership or non-membership in the Union, which is hereby recognized as a voluntary act on the part of the individual concerned.
- 3.1 The Union agrees that it will not interfere with, restrain or coerce the Employer in the performance of its management responsibilities.
- 3.2 No work which is normally and customarily performed by members of this Bargaining Unit shall be contracted out or contracted in, if as a result, any such member would be laid off, or have their regularly scheduled hours of work or hourly rate of pay reduced.

3.3 Discrimination

- (a) Each of the parties hereto agree that every person has a right to equal treatment with respect to terms of employment without discrimination because of race, ancestry, place of origin, colour, ethnic origin, citizenship, creed, sex, sexual orientation, age, record of offences, marital status, family status, or disability.
- (b) Every person who is an employee has the right to freedom from harassment in the workplace by the Employer or agent of the Employer, or by another employee, because of race, ancestry, place of origin, colour, ethnic origin, citizenship, creed, sex, sexual orientation, age, record of offences, marital status, family status, or disability.

3.4 Workplace Harassment

The Employer and the Union are committed to providing a harassment free workplace. The Ontario Human Rights Code defines harassment as "engaging in a course of vexatious comment or conduct that is known or ought reasonably be known to be unwelcome". Prohibited grounds for harassment in employment are listed in article 3.3 (b). Harassment may involve a threat, an implied threat or be perceived as a condition of employment.

All employees are expected to treat each other with courtesy, dignity and respect and to discourage harassment.

Properly discharged supervisory responsibilities, including the delegation of work assignments and/or the assessment of discipline that does not undermine the dignity of the individual is in no way to be construed as harassment.

If an employee believes they have been harassed and/or discriminated against on the basis of any prohibited ground of discrimination, there are specific actions that may be undertaken. The employee should request the harasser to stop the unwanted behaviour by informing the harassing individual(s) that the behaviour is unwanted and unwelcome. Should the employee not feel comfortable addressing the harasser directly, they may request assistance of the manager or a Union representative.

If the unwelcome behaviour was to continue, the employee should refer to the corporate harassment policy, which addresses informal and formal processes to deal with complaints. Employees are free to pursue all avenues, including the resolution/complaint procedures outlined within corporate policy.

An employee advancing a complaint may have the assistance of a representative of the Union or any other trusted person in any step of the process.

All employees will receive training in workplace harassment prevention and corporate policies to deal with harassment.

The Union is encouraged to submit, through the Home Administrator or directly to the Human Rights Division, suggestions for consideration in the development of workplace harassment training, policies and procedures.

3.5 Violence Against Women

The Employer and Union recognize and share the concern that women face situations of violence or abuse in their personal lives that may affect their attendance or performance at work.

There are other major traumatic events or medical conditions which may also affect an employee's attendance or performance at work.

When there is adequate verification from a recognized professional (i.e. doctor, professional counselor), indicating that an employee's performance or attendance would be affected by an abusive situation or other major condition or event, the Employer will give full consideration for the employee's circumstances in determining potential discipline or workplace accommodation measures within the terms of the collective agreement.

Such information will be treated in a confidential manner by the Employer and the Union unless required by law to be produced.

3.6 Staff Planning Consultation Committee

- (a) With respect to the development of any operational or restructuring plan which is likely to affect any of the Bargaining Units, the Union shall be involved in the planning process from the early phases through to the final phases of the process. Such involvement will be through the means of a Staff Planning Consultation Committee, to be comprised of equal numbers of representatives from the Corporation and the Union. The number of representatives shall consist of at least 2 from each Party, to a maximum of 6 from each Party. The Corporation shall make typing and other clerical assistance available as required. Each Party shall appoint a co-chair for the Committee, who shall alternate as meeting chairs. The respective Chairs will be jointly responsible for establishing the agenda of the committee meetings.
- (b) Once every three months or as otherwise agreed upon between the Parties, and wherever possible with 2 weeks advanced notice, a meeting of the Committee will be convened for the purposes of discussing the impact of such changes on the employment status of employees. Further, the participants will consider if practical ways and means exist to minimize any impact to affected employees, including:
 - identifying and proposing possible alternatives to any action that the Corporation may propose taking;
 - (ii) identifying and seeking ways to address retraining needs of employees;
 - (iii) identifying vacant positions within the Corporation for which the surplus members of the Bargaining Units may qualify, or such positions which are currently filled but which are expected to become vacant within a twelve (12) month period.

Where such practical ways and means cannot be developed, the employees affected will be subject to layoff in accordance with the terms of the Collective Agreement.

Meetings shall be held during normal office hours. Representatives attending such meetings during regularly scheduled hours of work shall not lose regular earnings as a result of such attendance.

Meetings shall operate by consensus. The Committee may submit recommendations to the Managing Director, Housing, Social Services and Dearness Home who will forward the recommendations. Nothing in the aforementioned shall preclude the Union from seeking delegate status to Standing Committees of Council.

To allow the meetings to proceed with the mandate, the Corporation shall provide participants with such pertinent and reasonable financial, staffing and available reorganization plans which directly affect the Bargaining Units, as long as such provision is not contrary to applicable law.

Any agreement between the Corporation and the Union resulting from the above review will take precedence over the provisions of the respective Agreements.

3.7 Technological Change

Where, as the result of reorganization, technological change or other operational change, new or greater skills are required than previously required, the Parties recognize the mutual responsibility both of affected employees and the employer to ensure that skills upgrading is undertaken. To the extent practical in consideration of operating efficiencies, the Employer will provide employees with opportunities to develop such skills within a reasonable time frame. The Employer will make reasonable efforts to develop a skills upgrading plan which may include external courses on the employee's own time.

ARTICLE 4 - COLLECTION OF UNION DUES OR UNION SECURITY

4.0 The Employer agrees that it will deduct union dues from the earnings of each employee coming within the scope of the bargaining unit defined in the recognition clauses of this agreement in accordance with the provisions of the Unifor Constitution in the manner and amounts provided as notified in writing by the union. These dues will be deducted bi-weekly and remitted monthly no later than 2 weeks after the final pay deposit of the month to the following address:

Unifor: Local 302 125 Elm Street London, Ontario

N5Z 2K4

Attention: Rusty Sproul

A list of employees, their phone numbers and their addresses for and on whose behalf such deductions have been made shall also be forwarded.

- 4.1 The Employer shall, when remitting such amount, provide the Union with the addresses of new employees on the first deduction along with a list of all employees from whom deductions have been made, stating the amount and reason for any change from the preceding month.
- 4.2 Such deductions, with respect to the new employees shall be made from the first regular deduction date following four weeks service (160 hours for Part-time employees).
- 4.3 The Employer agrees that a Union Representative shall be given the opportunity of interviewing each new employee once upon completion of the period identified in Article 4.2 for the purpose of informing such employee of the existence of the Union. The Employer shall advise the Union from time to time as to the names of the persons to be interviewed and the time and place of the interview, the duration of which shall not exceed fifteen (15) minutes. The interview shall take place on the Employer's premises, in a room designated by the Employer, and the employees shall report to this room for interview during the interview period.

ARTICLE 5 - NO STRIKES OR LOCKOUTS

5.0 The Parties acknowledge that the Home is a hospital within the meaning of the Hospital Labour Disputes Arbitration Act and they further agree that there shall be no strikes or lockouts as those expressions are defined in the Labour Relations Act.

ARTICLE 6 - RESERVATION OF EMPLOYER'S FUNCTIONS

- 6.0 The Union acknowledges that it is the exclusive function of the Employer to:
 - (a) Order, discipline and efficiently govern the conduct of employees, establish and enforce rules and regulations necessary therefore:

The Employer undertakes to give notice, wherever reasonable, of its intention to exercise its rights hereunder in a manner which will affect the working conditions of the employees.

- (b) Hire, discharge, transfer, lay off, promote, demote, classify or discipline employees, provided that a claim of discriminatory transfer, lay off, promotion, demotion or classification or a claim that an employee has been discharged or disciplined without a reasonable cause, may be the subject of a grievance, and dealt with as hereinafter provided.
- 6.1 The Employer reserves the right in all matters of scheduling to maintain adequate service levels, and it is recognized that provisions in the respective Agreements relating to scheduling matters will be administered accordingly.

ARTICLE 7 - REPRESENTATION

- 7.0 The Union acknowledges and agrees that the stewards and Union Committee members as described in Article 7.5 for Full –Time employees and Article 7.7 for Part Time employees, have regular duties to perform in connection with their employment and all activities will be carried on outside regular working hours, unless otherwise mutually agreed. The Union agrees that there will be no interference with any employment by reason of activities of the stewards or Union Committee members.
- 7.1 The Union agrees to supply the Employer with the names of the stewards and with the names of Union Committee members and will keep such list up to date at all times.
- 7.2 Union committee and Employer representatives will meet on a regular basis as mutually agreed to discuss labour management issues. Such meetings will take place no less than every two months. Such meetings will occur on the Employer's premises during normal business hours. Representatives attending such meetings during regularly scheduled hours of work shall not lose regular earnings as a result of such attendance. Employees scheduled to work an 8 hour evening shift or night shift on the day of the meeting (or day before for a night shift) will have the opportunity to have their shift rescheduled to accommodate the meeting.
- 7.3 The Union Chairperson shall be retained at work during any layoffs or cutbacks in employment during their term of office.
- 7.4 The Union shall have the right at any time to have the assistance of representatives of Unifor, for the purposes of joint meetings with management. A union representative of Unifor shall have reasonable access to the Home. Written permission must be obtained in advance for all other representatives.

Representation - Full-Time Employees Only

7.5 (a) The Employer acknowledges the right of the Union to appoint or otherwise select a Union Committee composed of not more than six (6) members, and the Employer will recognize the said Committee for the purpose of handling any grievance or bargaining on any matter properly arising from time to time, during the continuance of the Collective Agreement, including the negotiations for or renewal of any Agreement.

The above six (6) members of the Union Committee will be paid for attendance at negotiation meetings with Corporation managers to a maximum of eight hours per day, per Union Committee member, for all negotiation days up to and including the date of conciliation. It is specifically noted that in no circumstance will the payment for such negotiation attendance result in an overtime premium being paid during the work week, and that attendance at negotiations is considered to constitute the employee's shift for that day.

- (b) The Employer acknowledges the right of the Union to appoint or otherwise select sixteen (16) stewards.
- (c) At any meeting held to discuss a grievance the Union shall be entitled to representation by the Union Grievance Committee, consisting of three (3) members of the regular Union Committee and the steward representing the Department from which the grievance originated.

It is agreed that any two (2) of the Union Representatives of Unifor Local 302, and the National Representative of the area may be present with the Committee at the request of either the Union or the Employer.

7.6 The local chairperson will be permitted scheduled work time in accordance with Article 14 (paid at their regular wages) up to two (2) eight (8) hour shifts per calendar month to conduct union business. It is understood that such shifts will be day shifts. For the purpose of union business the day shift shall be 8:00 a.m. until 4:00 p.m. Such shifts shall be previously approved by the chairperson's manager or designate.

The parties agree and understand that any other scheduled work time that the local chairperson requests for union business and is approved by their manager or designate, which exceeds the two (2) eight (8) hour shifts per calendar month maximum will not be compensated.

Representation - Part-Time Employees Only

- 7.7 (a) The Employer acknowledges the right of the Union to appoint or otherwise select a Union Committee, who shall also act as Shop Stewards, composed of not more than four (4) members, and the Employer will recognize the said Committee for the purpose of handling any grievance or bargaining on any matter properly arising from time to time, during the continuance of the Collective Agreement, including negotiations for or renewal of any agreement. The Employer acknowledges the right of the Union to appoint or otherwise select seven (7) Stewards.
 - (b) The above four (4) members of the Union Committee will be paid for attendance at negotiations to a maximum of eight hours per Union Committee member, for all negotiations up to and including the date of conciliation. It is specifically noted that in no circumstance will the payment for such negotiation attendance result in an overtime premium being paid during the work week, and that attendance at negotiations is considered to constitute the employee's shift for that day.
 - (c) At any meeting to discuss a grievance, the Union shall be entitled to representation by the Union Grievance Committee, consisting of two Union Committee members and the griever concerned.
 - (d) It is agreed that a Union representative of Unifor Local 302, and/or the National Representative of the area, may be present with the Committee at the request of either the Union or the employer.

7.8 Health and Safety

Dearness Home

- (a) The parties agree that a Joint Health and Safety Committee shall be established and operate as required under the Ontario Occupational Health and Safety Act for the purpose of preventing accidents, injury and illness in the Home pertaining to the worker.
- (b) The Corporation shall take every precaution reasonable in the circumstances for the protection of a worker and provide a safe and healthy workplace, in accordance with the Ontario Occupational Health and Safety Act.
- (c) The Employer recognizes that the Joint Health and Safety Committee will be composed of three representatives from each bargaining unit for a total of six representatives, selected or appointed by the Union from amongst bargaining unit employees.

- (d) The meetings of the Joint Health and Safety Committee shall be held monthly as members are available, but as a minimum, at least every second month. The Committee shall maintain minutes of the meetings and approved minutes will be distributed by the Co-Chairs to the Home Administrator and Union.
- (e) The Home will provide information reasonably related to the Committee's activities and necessary to enable the Committee to fulfill its functions.
- (f) The Committee will review workplace accidents and illnesses, inspect the premises on a regular basis and report all findings for corrective action, investigate work related accidents and work as a group to resolve health and safety issues to maintain a safe workplace.
- (g) Terms of Reference for the Joint Health and Safety Committee will be maintained and revised to reflect any changes which are agreed to by the Committee, with the approval of the Home Administrator.
- (h) A member of the Committee is entitled to:
 - one hour or such longer period of time as the Committee determines is necessary to prepare for each Committee meeting;
 - (ii) such time as is necessary to attend meetings of the Committee; and
 - (iii) such time as is necessary to carry out the member's duties as outlined in the Ontario Occupational Health and Safety Act.

Payment will be in accordance with the terms of the collective agreement.

- (i) A national Unifor Health and Safety Representative may be invited by the Committee to attend Joint Health and Safety meetings.
- (j) Where possible, each year on April 28 at 11 a.m., employees are invited to pause and observe one minute of silence in memory of workers killed or injured on the job, unless such observance would place a resident's health or safety at risk.
- (k) Health and Safety training provided to employees covered by this collective agreement will be co-facilitated by a worker member of the Joint Health and Safety Committee and a management representative.
- (I) Both parties will notify each other in writing on an annual basis, or as changes occur, of the name of Committee members.

City Hall Cafeteria

A City Hall Cafeteria Joint Health and Safety Committee will be established in accordance with the Occupational Health and Safety Act. The Committee will be comprised of two (2) people, one member of Unifor and one member of management.

ARTICLE 8 - GRIEVANCE PROCEDURE

- 8.0 (a) No grievance shall be considered:
 - which usurps the function of the Management outside the scope of this Agreement.
 - (ii) where the circumstances giving rise to it occurred or originate more than ten (10) working days, excluding Saturday, Sunday and Statutory holidays, for Full-time Units or ten (10) calendar days, excluding Saturday, Sunday and Statutory holidays, for Part-time Units, before the filing of the grievance.
 - (b) Grievances properly arising under the Agreement shall be adjusted and settled as set out in this Article.

8.1 Complaint Resolution Step

It is the mutual desire of the Parties hereto that complaints of the employees shall be resolved as quickly as possible and it is understood that an employee has no formal grievance until they have first given to the immediate Management Supervisor concerned an opportunity of resolving the complaint. When an employee has a complaint, they (who may request the assistance and attendance of the steward) shall discuss it with the immediate Management Supervisor. The immediate Management Supervisor shall give a decision to the employee concerning their complaint within three (3) days, per article 8.3(c), and failing settlement, a grievance shall be dealt with in the following manner and sequence, provided it is presented within three (3) days, per article 8.3(c), following receipt of the decision from the Immediate Management Supervisor.

The immediate management supervisor shall contact a steward, or committee member, to be in attendance at the beginning of the meeting. The employee will be given the opportunity for the assistance and attendance of the steward during the meeting.

8.2 Grievance Procedure - Step 1

Within five (5) days, per article 8.3(c), after the decision is given by the immediate Management Supervisor, the grievance may be submitted in writing to the Home Administrator. It is understood that the Home Administrator (or designate) may have such counsel and assistance as they may desire, and that any two of the Union Representatives of Unifor Local 302 and the National Representative of the Union may also be present at the request of either the Employer or the Union. The Home Administrator (or designate) shall deliver their decision in writing to the employee within five (5) days, per article 8.3(c), after receiving the written grievance.

Failing settlement:

Grievance Procedure – Step 2

Within eight (8) days, per article 8.3(c), after the decision is given in Step No. 1, the grievance may be forwarded, in writing to the Director, People Services. A meeting between the aggrieved employee and the Union Grievance Committee and the Director, People Services or designate shall be held within eight (8) days, per article 8.3(c), following submission of the grievance, or as may be extended by mutual agreement of the Parties. It is understood that the Director, People Services may have such counsel and assistance as they may desire, or may elect to be represented by a designate, and that any two of the Union Representatives of Unifor Local 302 and the National Representative of the Union may also be present at the request of the Director, People Services, or the Union. The decision of the Director, People Services or designate shall be given in writing within eight (8) days, per article 8.3(c), following the meeting.

Referral to Arbitration – Step 3

Failing settlement of any grievance under the foregoing procedure arising from the interpretation, application, administration or alleged violation of the Agreement, including any question as to whether a matter is arbitrable, the grievance may be referred to Arbitration by either Party. If no written request for Arbitration is received within ten (10) days, per article 8.3(c), after the decision under Step 2 is given, it shall be deemed to have been settled.

- 8.3 (a) Should any grievance not be submitted within the various time limits specified under the Grievance Procedure, the Employer will not be obliged to consider it, provided that any of the time limits in Articles 8, 9 and 10 may be extended by mutual agreement of the Parties.
 - (b) All agreements reached under the Grievance Procedure between the representatives of the Union and the representatives of the Corporation,

will be final and binding upon the Corporation, the Union and the employees.

- (c) Any time limits referred to in the Grievance Procedure or Arbitration or any sub-section thereof within which any procedure is required to be taken or notice required to be given, shall be calculated exclusive of Saturday, Sunday, Paid Holidays (as declared) and, in the case of Fulltime units, an aggrieved employee's day off.
- (d) Where the differences arise between the Employer and the Union concerning the interpretation or violation of this Agreement, which may be considered as policy matters, the differences between the Parties shall be reduced to writing and entered into the Grievance Procedure in accordance with Article 11.
- (e) If the matter of the policy grievance is not satisfactorily settled, it is understood that it may be carried through the balance of the grievance procedure, including the step of arbitration for final and binding settlement upon the Parties.
- (f) Should two (2) or more employees have complaints which are the same or similar, such complaints, if submitted as a grievance, may be processed as a group grievance providing the parties to this agreement agree on a group grievance process at Step 1.

ARTICLE 9 - ARBITRATION

9.0 (a) When either Party requests that a grievance be submitted to Arbitration, the request shall be in writing addressed to the other Party of the grievance.

The parties shall appoint a single arbitrator from the following list:

- Brian Etherington
- Louisa Davies
- Chris White
- Jasbir Parmar

Arbitrators will be selected in order of the date of referral on a rotational basis unless the parties agree to consolidation. A list of a minimum of four names will be maintained during the term of this agreement. Revisions to the list will be mutually agreed upon.

If either party wishes to have the grievance submitted to a Board of Arbitration, they will so advise the other party within ten (10) working days of the notice referring the matter to arbitration. The sole arbitrator who would otherwise hear the grievance shall chair a three person Board of Arbitration. The parties shall appoint their nominee to the Board within ten (10) working days of the confirmation of the chair.

- (b) The said two nominees first appointed shall be at liberty, prior to the arbitration hearing, to discuss the grievance submitted to Arbitration with a view to the mutual settlement of the grievance so submitted by the Parties.
- (c) No matter may be submitted to Arbitration which has not been properly carried through all previous steps of the Grievance Procedure within the time limit in the manner provided.
- (d) No person may be appointed as an arbitrator who has been involved in an attempt to negotiate or settle the grievance.
- (e) Each of the Parties shall pay the expense of their own nominee and one half of the fees of the Arbitrator.
- (f) The sole Arbitrator or Board of Arbitration shall have no power to alter or change any of the provisions of this Agreement, or to substitute any new provision for existing provisions, nor to deal with any matter not covered by this Agreement.
- (g) The decision of any sole Arbitrator or Board of Arbitration shall be consistent with the terms and provisions of this Agreement.
- (h) Proceedings before the sole Arbitrator or Board of Arbitration shall be expedited by the Parties hereto. The decision of the sole Arbitrator or Board of Arbitration shall be final and binding on both Parties to this Agreement.
- (i) Any grievance involving the interpretation or application of this Agreement, which has been disposed of by the Arbitration process, shall not be made the subject of another grievance.
- (j) At any stage of the complaint or Grievance Procedure, including Arbitration, the Parties may have the assistance of the employee or the employees concerned as witnesses, and all reasonable arrangements will be made to permit the conferring Parties of the Board of Arbitration to have access to any part of the Home to view any working condition

- which may be relevant to the settlement of the grievance at a reasonable time, and so as not to interfere with the function of the Home.
- (k) The parties agree that the arbitration language in article 9 applies to rights arbitration only.

ARTICLE 10 - DISCHARGE CASES

- 10.1 (a) A claim by a permanent employee that they have been unjustly discharged shall be treated as a special grievance, if a written statement of such special grievance is lodged by the employee within five (5) days, per article 8.3(c), of the date of the notice of discharge.
 - (b) Such special grievance, to be considered, must be presented to the Home Administrator or designated representative by a member of the Union Committee. A special meeting between the Union Committee, Managing Director, Housing, Social Services and Dearness Home (or designate) and Director, People Services (or designate) will be held within five (5) days, per article 8.3(c), of receipt of such special grievance. Failing settlement of such special grievance under the foregoing procedure, the grievance may be referred to Arbitration, for final and binding settlement upon the Parties.
 - (c) The Employer shall notify the Chairperson in writing as soon as reasonably possible upon suspension or termination of any employee of the Bargaining Unit.
- 10.2 The Board of Arbitration or the single Arbitrator, as the case may be, may deal with such special grievance by:
 - (a) confirming the Employer's action in dismissing the employee.
 - (b) re-instating the employee with or without compensation for time lost; or
 - (c) in any other manner as is deemed just or equitable in the opinion of the Board of Arbitration or the single Arbitrator.

ARTICLE 11 - POLICY GRIEVANCES

- 11.0 Either the Union or the Employer may initiate a policy grievance. A policy grievance shall be in writing. A policy grievance of the Employer shall be delivered to the Chairperson of the Union or to a Union Representative or International representative of the Union.
- 11.1 A policy grievance of the Union shall be commenced at Step 2 of the Grievance Procedure. Any policy grievance shall be filed within eight (8) days of the incident

giving rise to the grievance. If any such grievance is not settled within eight (8) days of its delivery pursuant to this paragraph, the matter may be referred to Arbitration under Article 9.

11.2 Policy Grievance defined as:

A grievance which alleges an actual violation of a specific provision of this Agreement and which could not otherwise be resolved at lower steps of the Grievance Procedure because of the nature or scope of the subject matter of the grievance. Such grievance is to be submitted at Step 2 of the Grievance Procedure.

ARTICLE 12 - SENIORITY

- 12.0 Seniority for employees in each of the bargaining units is as more particularly described in Articles 12.3, 12.4 and 12.5 for Full Time employees and Articles 12.6, 12.7 12.8 for Part Time employees.
- 12.1 Seniority based on date of hire shall govern in all cases of promotion, demotion, transfer to a higher paid job or to a job with equal pay. The more senior employee must be qualified to do the job involved efficiently and in the case of demotion they must be capable of efficiently performing the duties of the job to which they are demoted. For purposes of vacation scheduling only and in circumstances involving a former part-time employee who has the identical length of service as an employee who has been a full-time employee for all service with the Employer, the full-time employee will be deemed to have greater seniority.

Where two or more employees commence work for the Employer on the same day, the precedence in position on the seniority list will be determined by lottery. The Union Chairperson or designate and all affected employees must be present for the lottery, numbers will be drawn and sub-seniority dates established. The affected employees will sign a letter acknowledging their sub-seniority date.

12.2 An employee who accepts a temporary assignment in a bargaining unit ("temporary") at the Dearness Home other than the bargaining unit in which the employee currently has seniority ("permanent") will continue to accumulate seniority in the permanent bargaining unit during the temporary assignment. Seniority in the permanent bargaining unit will apply in the temporary bargaining unit for all purposes except for job posting and layoff, and except where otherwise noted such employees will be governed by the terms of the Collective Agreement of the temporary bargaining unit during the temporary assignment.

Seniority - Full-Time Employees Only

- 12.3 Seniority is defined as length of continuous service since the date of last hire by the Employer and will be acquired when an employee has completed 45 days worked. Such seniority will date from the first day that an employee actually commenced work for the Employer within the bargaining unit, and will accumulate thereafter. All employees will be regarded as probationary employees until they have acquired seniority as above provided, although an employee shall be entitled to the assistance of the Union in settling a grievance other than dismissal in accordance with the Grievance Procedure herein set forth, after a period of thirty (30) days worked for Full-time employees employment. The dismissal of a probationary employee shall not be the subject of a grievance.
- 12.4 The probationary period may be extended by thirty (30) working days in extenuating circumstances as determined by the Home Administrator (or designate). The Home Administrator (or designate) will advise the Unit Chairperson of the extension and the reason for such extension.
- 12.5 When a permanent employee is transferred to another classification within the Bargaining Unit, they shall be subject to a trial period in their new duties of thirty (30) working days. They shall receive the starting rate in the new classification for this trial period of thirty (30) working days at the end of which time, if they continues in the new duties, they shall move to the same Schedule "A"; wage grid level in the new position as enjoyed in immediate previous position. If such employee wishes to return to their former position, they may do so by making written request to the Home Administrator (or designate) during the trial period. In addition, if the employee fails to satisfactorily complete the trial period, or, is unable to perform the work, they shall, subject to the right of grievance by either Party, be returned to their previous position and wage rate. Similarly, any other employee who has subsequently been promoted or transferred as a result of the original promotion will return to their previous position and salary.

Seniority - Part - Time Employees Only

12.6 Seniority is defined as length of continuous service since the date of last hire by the Employer and will be acquired when an employee has completed 360 hours for Part-time employees continuous service. Such seniority will date from the first day that an employee actually commenced work for the Employer, and will accumulate thereafter. All employees will be regarded as probationary employees until they have acquired seniority as above provided, although an employee shall be entitled to the assistance of the Union in settling a grievance other than dismissal in accordance with the Grievance Procedure herein set forth, after a period of 220 hours for Part-time employees employment. The dismissal of a probationary employee shall not be the subject of a grievance.

- 12.7 The probationary period may be extended by 220 hours in extenuating circumstances as determined by the Home Administrator (or designate) as applicable. The Home Administrator (or designate) will advise the Unit Chairperson of the extension and the reason for such extension.
- 12.8 When a Part-time employee is transferred to another classification within the Bargaining Unit, they shall be subject to a trial period in their new duties of two hundred and forty (240) working hours. They shall receive the starting rate in the new classification for this trial period of 240 working hours at the end of which time, if they continue in the new duties, they shall move to the same Schedule "A"; wage grid level in the new position as earned in immediate previous position. If such employee wishes to return to their former position, they may do so by making written request to the Home Administrator (or designate) during the trial period. In addition, if the employee fails to satisfactorily complete the trial period, or, is unable to perform the work, they shall, subject to the right of grievance by either Party, be returned to their previous position and wage rate. Similarly, any other employee who has subsequently been promoted or transferred as a result of the original promotion will return to their previous position and salary.

12.9 Layoff and Recall

- (a) There shall be at least 6 months notice to the Union in the event of a proposed layoff of a permanent or long-term nature or in the event of a substantial bed cutback which affects or could affect the bargaining unit, unless the notice of reduced funding from the Ministry to the Home is given less than 6 months from the end of the fiscal year, in which case the number of months notice required shall be that number of months from the end of the fiscal year that the notice from the Ministry was received but in no case shall be less than 3 months.
- (b) In the event of a layoff of a permanent or long-term nature, the Corporation will provide affected employees with 2 weeks notice for each year of service to a maximum of 16 weeks, provided that the employee has more than 12 months service. Employees with less than 12 months service will be entitled to notice in accordance with the provisions of the Employment Standards Act. A copy of any notice of layoff to an employee will be provided to the Union at the same time.
- (c) In the event of a layoff, the Corporation shall layoff employees in the reverse order of their seniority within their classification in the respective bargaining unit i.e., full-time or part-time, providing that there remain on the job employees who then have the ability to perform the work.

- (d) An employee who is subject to layoff shall have the right to either:
 - (i) accept the layoff, or
 - (ii) displace an employee who has less bargaining unit seniority and who is the least senior employee in a lower or identical paying classification in the bargaining unit if the employee originally subject to the layoff can perform the duties of the lower or identical classification without training other than orientation. Such employee so displaced shall be laid off subject to the layoff procedure. A full time employee may displace a part time employee and vice versa, given reasonable orientation.

Note - an identical paying classification shall include any classification where the straight time hourly wage rate at the level of service corresponding to that of the laid off employee is within 1% of the laid off employee's straight time hourly wage rate.

In the event that there are no employees with lesser seniority in lower or identical paying classifications as defined in this Article, a laid off employee will have the right to displace an employee with less seniority, who is the least senior employee in a classification where the straight time hourly rate at the level of service corresponding to that of the laid off employee is within 5% of the laid off employee's straight time hourly rated provided they can perform the duties without training other than orientation. Such employee so displaced shall be laid off subject to the layoff procedure.

- (e) An employee shall have opportunity of recall from a layoff to an available opening, in order of seniority, provided they possess the qualification and ability to perform the work before such opening is filled on a regular basis under the job posting procedure. The posting procedure shall not apply until the recall process has been completed.
- (f) In determining the qualification and ability of an employee to perform the work for the purposes of the paragraphs above, the Corporation shall not act in an arbitrary or unfair manner.
- (g) An employee recalled to work in a different classification from which they were laid off shall have the privilege of returning to the position they held within two years of having been recalled.
- (h) No new employee shall be hired until all those laid off have been given an opportunity to return to work and have failed to do so, in accordance

- with the loss of seniority provisions, or have been found unable to perform the work available.
- (i) Employees on layoff or notice of layoff shall be given preference for temporary vacancies which are expected to exceed 10 working days. An employee who has been recalled to such temporary vacancy shall not be required to accept such recall and may instead remain on layoff.
- (j) No full-time employee shall be laid off by reason of their duties being assigned to one or more part-time employees.
- (k) In the event of a layoff commenced on the day immediately following a paid holiday, an employee otherwise qualified for holiday pay shall not be disentitled thereto solely because of the day on which the layoff commenced
- (I) In the event of layoff of a full-time employee, the Corporation shall pay its share of insured benefits premiums up to 3 months from the end of the month in which the layoff occurs or until the laid off employee is employed elsewhere, whichever comes first.
- 12.10 Seniority lists will be posted on the official Union Bulletin Board and a copy of the Seniority List will be supplied to the Union Office and the Chairs of the Union. The Seniority List will be posted and revised semi-annually on or before January 31st and July 31st of each year according to the records of the Corporation. Seniority as posted will be deemed to be final and binding and not subject to complaint unless such complaint is made within thirty (30) days from the first date of current posting.
- 12.11 An employee shall lose all seniority and shall be deemed to have quit the employ of the Corporation after they:
 - (a) are discharged for cause and the discharge is not revoked through the Grievance Procedure;
 - (b) tender their resignation in writing or quits the employ of the Employer;
 - (c) fail to report for work within 10 days after being notified by the Employer following a lay off or fails to advise the Employer within five (5) days (exclusive of Saturdays, Sundays or paid holidays) of their intention to report for work, pursuant to notification by registered mail addressed to the last address on record with the Home. The notification shall state the job to which the employee is eligible to be recalled and the date and time at which the employee shall report to work. The employee is solely responsible for their proper address being on record with the Home;

- (d) are absent for five consecutive regularly scheduled shifts without providing proper notice of a reason satisfactory to the Employer. Any employee who is terminated in accordance with this provision will be advised by registered mail or equivalent and a copy of this notice will be sent to the Chairperson of the Bargaining Unit at the same time;
- (e) are laid off continuously for 30 months or longer.
- 12.12 Where a Full-time employee transfers to the Part-time unit, or vice versa, in the same job classification, there will be no trial period. The employee will transfer at their current wage rate.

ARTICLE 13 - JOB CLASSIFICATION AND RATES OF PAY

- 13.0 Job classifications and rates of pay shall be as set out in Schedule "A" to this Agreement.
- 13.1 When an employee relieves in a higher paid classification for one (1) hour or more, they shall be paid at the next higher rate in the higher classification for all hours worked in such higher classification.
- When a new classification which is to be included in this Agreement is established by the Employer, the Employer shall provide the Union with a copy of the new Job Description and shall consult with the Union and subsequently determine a rate of pay. In the event that the Union disagrees with the rate so established, it may lodge a policy grievance and the provisions of Article 11 of this Collective Agreement shall apply.

ARTICLE 14 - HOURS OF WORK

- 14.0 (a) The hours of work for the various units shall be as more particularly described in Articles 14.8 and 14.9 for Full –Time employees and Articles 14.10 and 14.11 for Part –Time employees.
 - (b) Employees working in Departments which operate on a twenty-four (24) hour basis shall be required to report for duty at their respective stations five (5) minutes before their respective shift begins.
- 14.1 Employees shall not be required to work more than five (5) days consecutively except in the Engineering Section where it shall be not more than seven (7) days consecutively. For Full-time employees the Employer will provide the employees one weekend off (Saturday and Sunday) in every two (2) weeks. It is understood that a day is defined as that on which the majority of hours is worked except when

an employee in the RPN classification is working a night shift on a Friday, in such cases they will be considered to have worked on the Friday.

- 14.2 (a) An employee who works a majority of hours on the afternoon or night shift shall receive a \$0.70 shift premium per hour payable for all hours worked.
 - (b) Schedules of work shifts on a minimum four week (Sunday to Saturday) turnabout basis shall be posted at least two weeks in advance of the current work period and remain posted for the duration of the work period schedules.
 - (c) In the event an employee reports for work on a regularly scheduled shift and is sent home for lack of work, the employee will be guaranteed either three (3) hours work or a minimum of three (3) hours pay at their regular rate.
- 14.3 Where any employee covered by this Agreement is required to work ten (10) consecutive hours or more, they shall be entitled to a meal allowance of eight dollars (\$8.00).
- 14.4 When a conversion from Standard Time to Daylight Saving Time, or Daylight saving Time to Standard Time, occurs, employees working the duty shift during which the conversion occurs will be paid for all hours actually worked. Where the number of hours actually worked exceeds eight (8), those hours in excess will be paid at the appropriate overtime rate.
- An employee may request in writing to exchange a shift of equal duration or an eight (8) hour shift with a four (4) hour shift with an equally qualified employee in the same classification provided there is no additional cost for the service being provided and the permission for exchange is at the discretion of the Manager or designate.
- 14.6 Employees must request vacation or other absent time for Christmas and New Year period no later than October 1st in each year. The Employer will draw up and post the schedule for the 8 weeks ending with the week including the two consecutive days following New Year's no later than November 1st in each year; in the event that an employee does not request a schedule by October 1st, they will forfeit the right of selection.
- 14.7 A weekend shift premium of \$0.35 (effective November 14, 2019, \$0.45; effective July 1, 2020, \$0.55) will be paid on all hours worked between Friday 10:00 p.m. and Sunday 10:00 p.m.

14.8 Hours of Work – Full-Time Employees Only

- (a) The standard work week shall be forty (40) hours of five (5) days, on an eight (8) hour shift basis, with one fifteen (15) minute rest period and one twenty (20) minute lunch period.
- (b) Employees called in at a time other than their regular shift shall be paid for at least three (3) hours of work at their overtime rate.
- (c) The Employer may designate employees to be on standby duty. While designated as on standby, the employee shall be paid two dollars (\$2.00) per hour for all hours of such duty, even if during such duty, they are recalled to work and paid the appropriate overtime rate. The employee must remain available for recall at all times while on standby duty.
- All overtime work in excess of the regularly scheduled shift and /or eighty (80) hours in the bi-weekly pay period shall be paid for at the rate of time and one-half, computed at the employee's regular rate of pay. It is understood that all overtime must be authorized by the immediate management supervisor. Notwithstanding the foregoing, time worked in excess of the regular scheduled shift on a Statutory Holiday shall be paid at double time computed at the employee's regular rate of pay.

14.10 Hours of Work - Part - Time Employees Only

- (a) The hours of work will be as scheduled by the Home. The hours of work will be no more than 8 hours and no less than 4 hours. When the daily hours of work are eight (8) hours or greater than five (5) hours, they will be inclusive of one twenty-minute lunch period and one fifteen-minute rest period during the shift. Where the daily hours of work are five (5) hours or less, they will be inclusive of one rest period of fifteen (15) minutes. Shifts would be scheduled at the discretion of the Home Administrator, taking into consideration the needs of the residents. No scheduled shift shall be less than 4 hours unless mutually agreed by the parties.
- (b) Employees shall not be required to work more than five (5) days consecutively except in the Engineering Section where it shall be not more than seven (7) days consecutively.
- (c) Scheduled work will be offered on as equitable a basis as possible between employees in the bargaining unit.

- (d) Any part-time employee covered by this Collective Agreement, who may be required from time to time to work in excess of 48 hours in a bi-weekly pay period shall continue to be covered by this Collective Agreement.
- (e) Where an employee is requested to work a shift, and such request does not allow the employee one hour notice prior to the start of the shift, provided the employee reports for duty not more than one hour after the scheduled start of the shift, and works the balance of the shift, the employee will be paid for all hours in the scheduled shift. Under such circumstances, the employees will endeavour to report promptly.
- (f) The Employer will provide part –time employees one (1) weekend off (Saturday and Sunday) in every two (2) weeks, unless otherwise mutually agreed between the employee and employer. It is understood that a day is defined as that on which the majority of hours is worked except when an employee in the RPN classification is working a night shift on a Friday, in such cases they will be considered to have worked on the Friday.
- (g) In the event an employee (full-time or part-time) is unable to work an eight (8) hour scheduled shift and the Employer becomes aware of their unavailability prior to completion of the schedule of work shifts referred to in Article 14.1, regular part-time employees in the same classification who are scheduled to work a four (4) hour shift which falls within the first or last four (4) hours of that eight (8) hour shift shall be offered the opportunity to work the additional four (4) hours of the shift in order of seniority before other employees are called in or scheduled to work on an overtime basis. Such opportunities will only be offered provided they do not result in any overtime or other costs to the Corporation that would not otherwise be incurred
- 14.11 All overtime work in excess of the regularly scheduled shift (defined as 8 (eight) hours for this Article) and/or eighty (80) hours in the bi-weekly pay period shall be paid for at the rate of time and one-half, computed at the employee's regular rate of pay. It is understood that all overtime must be authorized by the immediate management supervisor. Notwithstanding the foregoing, time worked in excess of the regular scheduled shift on a Statutory Holiday shall be paid at double time computed at the employee's regular rate of pay.
- 14.12 (a) Notwithstanding Article 14.11 which provides for overtime to be paid when earned, the Employer agrees to allow employees to accumulate overtime balances to be paid out any time prior to the end of the calendar year in which it is earned.

- (b) A maximum of forty (40) hours may be carried over into the next year. Any remaining overtime hours will be paid out on the last pay period of the year.
- (c) Overtime can be requested to be paid out at any time during the year provided that it has been requested on the appropriate form and has been submitted before the end of any pay period.

ARTICLE 15 - PAID HOLIDAYS

- 15.0 Paid Holiday provisions shall be as more particularly described in Article 15.2 for Full Time employees and Article 15.3 for Part Time employees.
- 15.1 Every employee, except those in the Engineering Section, shall be entitled to at least 3 consecutive days off at either Christmas or New Year's. Such days off shall be allocated on a classification seniority choice basis. It is understood and agreed that all normal working schedules extending over the Christmas and New Year period in each calendar year, shall be discontinued. It is understood that the practice of granting at least three consecutive days off for part-time employees is subject to the need to maintain adequate coverage for all Sections.

Subject to article 6.1, when a surplus of staff exists after all requests have been considered, the Employer will schedule both Christmas and New Years off by seniority.

The Lieu Day is to be taken within the year it is accumulated. In the event the employee is unable to take the lieu day, it will be paid out at the end of the year.

15.2 Paid Holidays - Full - Time Employees Only

(a) The following shall be considered as Paid and Civic Holidays:

New Year's Day

Labour Day

Good Friday Easter Monday Thanksgiving Day Remembrance Day

Victoria Day

Christmas Day

Canada Day

Boxing Day (December 26)

Civic Holiday

Lieu Day

Family Day

and any other day declared by a competent authority to be a holiday within the meaning of the Employment Standards Act.

- (b) Employees required to work on a Paid Holiday shall at the option of the Employer, receive either:
 - (i) Pay at the rate of time and one-half of the employee's regular rate for work performed on such holiday in addition to the employee's regular pay; or
 - (ii) Pay at the rate of time and one-half the employee's regular rate for work performed on such holiday and the employee shall receive an alternate day off with pay as scheduled by the Employer to occur within 90 calendar days before or following the Paid Holiday.
- (c) Where a Paid Holiday falls on a day which is not a regular work day for an employee, the employee shall receive an alternate day off with pay scheduled by the Employer to occur within 90 calendar days before or following the Paid Holiday.
- (d) In order to qualify for a payment of Paid Holidays an employee must work their regular working day immediately prior to the holiday and their regular working day immediately following the holiday, unless absent due to illness (for a period of twelve (12) consecutive months or less) or other reasonable excuse, where satisfactory evidence has been provided. Employees, who are absent due to illness for a period in excess of twelve (12) consecutive months, will not qualify for a payment of Paid Holidays.
- (e) Where the majority of the hours in an employee's regular shift falls within the hours of 12:01 a.m. and 12:00 midnight on a Paid Holiday, the entire shift shall be classified as Paid Holiday work. Where a minority of the hours in an employee's regular shift falls within the hours of 12:01 a.m. and 12:00 midnight on a Paid Holiday, the entire shift shall be classified as regular shift work.

15.3 Paid Holidays – Part – Time Employees Only

The provisions of this article shall apply only to those employees in this Bargaining Unit.

- (a) Employees are entitled to receive 8 (eight) hours pay on each of the under noted days, provided:
 - the employee has been employed for more than three calendar months;
 - the employee earned wages on at least eight days of the four work weeks, or worked at least 64 hours, preceding the holiday;
 - the employee has worked on their scheduled day of work, both preceding and following the holiday;

 the employee, having agreed to work on the public holiday, actually reports for, and performs, the required work for the full shift:

New Year's Day
Good Friday
Easter Monday
Victoria Day

Labour Day
Thanksgiving Day
Remembrance Day
Christmas Day

Canada Day Boxing Day (December 26)

Civic Holiday Family Day

and any other day declared by a competent authority to be a holiday within the meaning of the Employment Standards Act.

- In addition to the above, an employee may elect to be paid, if qualified in accordance with the above for one additional day to be known as a Lieu Day. The Lieu Day will be scheduled by mutual agreement between the employee and their immediate management supervisor.
- (b) Employees not covered in a) will receive pay for each of the above noted public holidays according to the Employment Standards Act.
- (c) Employees required to work on a Paid Holiday shall receive pay at the rate of time and one-half the employee's regular rate of pay performed on such holiday in addition to the employee's regular pay.
- (d) Where the majority of the hours in an employee's regular shift falls within the hours of 12:01 a.m. and 12:00 midnight on a Paid Holiday, the entire shift shall be classified as Paid Holiday work. Where a minority of the hours in an employee's shift falls within the hours of 12:01 a.m. and 12:00 midnight on a Paid Holiday, the entire shift shall be classified as regular shift work.
- (e) Subject to the need for maintaining adequate service coverage for all Sections as determined by the Home Administrator, consideration will be given upon written request made at least one month in advance, to employees in order of seniority by classification governed by this Collective Agreement, to being given either Christmas Day or New Year's Day each year as a day upon which work will not be scheduled for such employee, it being understood the request will be for only one of the two days.
- (f) Employees scheduled to work Christmas Day or New Year's Day will be given at least four weeks' notice of such scheduling.

ARTICLE 16 - SICK LEAVE

- 16.0 Sick Leave provisions shall be as more particularly described in Article 16.3 for Full-Time employees and Article 19.2 for Part Time employees.
- 16.1 (a) In the interest of facilitating rehabilitation at work of ill and injured employees, the parties will make every reasonable effort to return employees to modified duty positions. The Union will be notified in writing of such rehabilitation situations and the estimated length of the modified duties assignment as soon as possible. An employee who is ill or injured and unable to perform their regular duties may displace the junior position in a classification but only for the duration of the disability following which the disabled employee and the displaced employee may return to their original classification. The parties agree to meet their obligations to cooperate under the Human Rights Code when returning employees to modified work. In meetings regarding return to work plans employees will be offered union representation. Employees on rehabilitative assignments and related displaced employees will be eligible to apply for any posted vacancy.

The Return to Work Committee shall consist of three (3) Union and three (3) Management representatives for the purpose of returning those employees from the bargaining unit with occupational or non-occupational disabilities or diminished capacity to gainful employment, with the main objective to return those employees to their regular pre-disability work. The Return to Work Committee will meet monthly if necessary.

- (b) Employees shall make themselves available for any alternate or modified work which is within their abilities as supported by medical documentation satisfactory to the Director, People Services or designate and is work which the Employer determines is available.
- (c) Employees who are absent from work due to compensable injury shall, where such absences continue beyond 21 calendar days, advise the Employer of their health status, whether they would be able to assume alternate or modified duties, and a prognosis for their return to full duties, and shall provide similar advice at the conclusion of each subsequent 21 day period of continuing absence. In extenuating circumstances involving major and long-term disabilities the management may waive the requirement to provide written documentation. Employees shall make themselves available for any alternate or modified work which is within their abilities as supported by medical documentation satisfactory to the Employers' Health service provider and is work which the Employer determines is available.

- (d) When an employee will be unable to report for work because of sickness or other personal emergency reason, they shall, not later than one (1) hour on the day shift but two (2) hours on any other shift, before they are scheduled to be at work, inform their Supervisor or other management person in charge that they will not be at work, and shall further inform the Employer of the probable length of the absence in the event it is anticipated to be more than one (1) day, unless circumstances giving rise to the absence were beyond the control of the employee.
- An employee who is injured during their scheduled shift and who is required to leave for treatment or is sent home as a result of such injury shall receive payment for the balance of the shift at their regular rate of pay. Such employee shall be provided with transportation to a doctor's office or hospital and/or to their residence when unfit to drive.

16.3 Sick Leave – Full – Time Employees Only

- (a) Effective January 1, 1988, each permanent employee with a seniority date prior to January 1, 1988 shall be eligible to a one day sick leave credit for each month of service with the Corporation. Such credits shall be cumulative.
- (b) Each permanent employee with a seniority date on or after January 1, 1988 shall earn (3/4) three quarters, of a day (6 hours) of sick leave credit for each complete month of service with the employer (effective January 1, 2001). Vacation, Paid holidays, bereavement leave, time off in lieu or authorized Union business are considered hours worked. Such credits shall be cumulative to a maximum of 80 (eighty) days. Those employees with a seniority date of January 1, 1988 or later with a balance in excess of 80 days may use the present credits until they reach 80 days, at which point they will no longer earn in excess of 80 days (effective January 1, 2001).
- (c) The employer will provide a "top-up" to Workplace Safety and Insurance Board benefits which are paid to an employee for Temporary Total Disability. The "top-up" amount will be equal to the difference between the aforementioned W.S.I.B. benefit and 100% of the employee's net regular pay. The net regular pay is defined as that net amount regularly paid to an employee after the statutory deductions for Income Tax, Canada Pension Plan premiums and Employment Insurance premiums are deducted from the gross regular pay of an employee.
- (d) If an employee is absent due to sickness:

- (i) For three (3) but less than thirty (30) consecutive days, they shall be required to submit a certificate from their physician in order to be entitled to Sick Leave with pay which certificate is also required to state that they are able to resume full normal duties or indicate if there will be restrictions and duration of restrictions, on their return to work.
- (ii) For thirty (30) or more than thirty (30) consecutive days, they shall be required to submit a certificate from their physician certifying their inability to work and the nature of such sickness. An employee shall not be entitled to be paid Sick Leave in that month without furnishing such certificate and so on from month to month in the event the employee's sickness extends from one month to the next month.
- (iii) Chiropractor's certificates supporting an employee's inability to attend work will be accepted by the Home management if recognized by competent authorities such as Ontario Ministry of Health.
- (e) Subject to the provisions of the Municipal Act, 2001 (as may be amended from time to time), every employee with ten (10) or more years' service, who is actively engaged at their duties, may be granted Retirement leave, with full pay, at the discretion of the Director, People Services, for a period equal to the unexpended portion of their accrued Sick Leave credit, but not in excess of six (6) months, and according to the following schedule:

	Service Requirements	Period of Leave Up to:	
(i)	At least ten (10) and not more than fifteen (15) years	Three (3) calendar months	
(ii)	At least fifteen (15) and not more than twenty (20) years	Four (4) calendar months	
(iii)	At least twenty (20) and not more than twenty-five (25) years		
(iv)	More than twenty-five (25) years	Six (6) calendar months	

Employees with a seniority date on or after January 1, 1988 are not entitled to a sick leave gratuity as provided in Article 16.3 (e), (f) and (g).

(f) Subject to the provisions of the Municipal Act, in case of the death of an employee with ten (10) or more than ten (10) years of service, their estate shall be entitled to a portion of the balance of their accumulated Sick Leave Credits, if any, as follows:

Portion of Sick Leave credits to which employee is entitled, up to maximum shown:		
Ten (10)	Fifty percent (50%) up to ninety (90) days	
Eleven (11)	Fifty-five percent (55%) up to ninety-nine (99) days	
Twelve (12)	Sixty percent (60%) up to one hundred and eight (108) days	
Thirteen (13)	Sixty-five percent (65%) up to one hundred and seventeen (117) days	
Fourteen (14)	Seventy percent (70%) up to one hundred and twenty-six (126) days	
Fifteen (15)	Seventy-five percent (75%) up to one hundred and thirty-five (135) days	
Sixteen (16)	Eighty percent (80%) up to one hundred and forty-four (144) days	
Seventeen (17)	Eighty-five percent (85%) up to one hundred and fifty-three (153) days	
Eighteen (18)	Ninety percent (90%) up to one hundred and sixty-three (163) days	
Nineteen (19)	Ninety-five percent (95%) up to one hundred and seventy- one (171) days	
Twenty (20) or more	One hundred percent (100%) up to one hundred and eighty (180) days	

Employees with a seniority date on or after January 1, 1988 are not entitled to a sick leave gratuity as provided in Article 16.3 (e), (f) and (g).

(g) In case of termination of employment after five (5) years of service, an employee shall be entitled to a portion of the balance of their accumulated Sick Leave Credits, if any, as follows:

Portion of sick leave credits to which employee is entitled, up to maximum Years of Service shown:

- Five (5) or more but less than fifteen (15) years of service Fifty percent (50%) up to ninety (90) days
- Fifteen (15) or more but less than twenty (20) Fifty percent (50%) up to one hundred and twenty (120) days

Employees with a seniority date on or after January 1, 1988 are not entitled to a sick leave gratuity as provided in Article 16.3 (e), (f) and (g).

(h) If an employee has had three (3) periods of Sick Leave with pay in a calendar year, the Employer may withhold payment of sick leave for any subsequent absences in that year, notwithstanding that the employee may have accumulated Sick Leave Credits. It is the policy of the Employer and the Union to prevent abuse of Sick Leave and the withholding of any sick leave payment under this Article shall be in addition to and not in substitution for any other action the Employer may take in the event of an unauthorized absence of an employee. Nothing in this clause shall prevent an employee from filing a grievance if they are denied Sick Leave with pay or are disciplined or discharged for alleged unauthorized absence.

ARTICLE 17 - VACATIONS

17.0 Full-Time and Part-Time Employees

(a) Vacations with pay are granted for past services.

There will be two (2) vacation periods per year:

Vacation Period #1 - covering the period April 1 to October 31

It shall be the duty of the Employer to distribute and receive from each employee a "request form for vacation entitlement" from January 15 through February 15, of each year.

The scheduled vacation allotments will be posted for Vacation Period #1 no later than April 1 in each year.

Vacation Period #2 - covering the period November 1 to March 31

It shall be the duty of the Employer to distribute and receive from each employee a "request form for vacation entitlement" between August 15 – September 15 of each year.

The scheduled vacation allotments will be posted for Vacation Period #2 no later than November 1 in each year.

Any employee who fails to complete and submit the request form within the timeframes noted above for Vacation Period #2, shall be assigned vacation dates without regard to seniority.

The Employer will then arrange suitable dates for vacation allotments for each employee, with such allotments to be governed by overall seniority within classifications coupled with the Employer's right to maintain adequate service within all sections. The Employer will make every reasonable effort to ensure that the provisions for distributions of vacation by seniority and classification are observed.

Note: In exercising the seniority privilege, it is restricted to the time of vacation selection as specified in this Article.

- (b) Employees should normally take their vacation all at one time: those with more than three (3) years' service to take at least two (2) weeks at one time and one week later on, if section business requires the time to be divided. Notwithstanding the preceding requirement, the parties agree that an employee may request up to two (2) weeks' vacation as single days. All requests for single vacation days must be submitted in writing to the Employer at least ten (10) days in advance of the requested day off. The employer will have three (3) days (excluding Saturday, Sunday and Holidays) to notify the employee if the vacation was granted or denied. Such requests will not be unreasonably denied and will be subject to the Employer's right to maintain adequate service within all sections.
- (c) Where a Paid Holiday falls within an employee's vacation period, it shall be added to the vacation.
- (d) Employees absent from the workplace, in receipt of Long Term Disability or Workplace Safety and Insurance Board benefits, for greater than twelve (12) consecutive months will not continue to earn vacation credits until such time as they return to work for the Corporation.

17.1 Vacation - Full-Time Employees Only

(a) An employee shall be credited with one day's vacation for each month or major portion thereof of service in the first vacation year. (The vacation year shall extend from April 1 to March 31 next following.)

- (b) An employee with one year of service or more as of March 31 in any year shall be entitled to three (3) weeks' vacation.
- (c) An employee who has continuous past service of seven (7) years or more as of March 31 in any year, shall be entitled to four (4) weeks' vacation.
- (d) An employee who has at least 15 years of continuous past service as of March 31 in any year shall be entitled to five (5) weeks of vacation.
- (e) An employee who has continuous past service of twenty-five (25) years or more as of March 31 in any year shall be entitled to six (6) weeks' vacation.
- (f) All vacations must be taken during the year in which they are allowed, and may be taken at any time during the year, subject to the right of the immediate Management Supervisor for good and sufficient reason to vary or postpone, and subject to the provisions of Article 17.1 (g) hereof.
- (g) Any employee who foregoes their vacation in the current vacation year, at the request of their supervisor, for good and sufficient reason, may receive such postponed vacation during the first six months of the next following year, but such postponement shall be reported by the Home Administrator (or designate) to the Director, People Services for record keeping purposes.
- (h) An employee, after four (4) years continuous service may, upon request to the Immediate Management Supervisor given on or before January 1 in any year, by specifying definitely their reason in writing for desiring accumulation of vacations, such reason to be satisfactory to the Immediate Management Supervisor, forego one week's vacation for each year so that the employee may add to their credit a period not exceeding six (6) weeks' vacation at one time. The Home Administrator (or designate) shall report same to the Director, People Services for record keeping purposes. Any employee failing to use accumulated vacation for the reason specified in their application shall forfeit their accumulated vacation.
- (i) Any employee voluntarily terminating their employment shall receive pay for the vacation period to their credit, at the time of the termination of their employment, in addition to their regular pay.
- (j) No vacation time shall be lost:
 - as the result of an accident or illness covered by an award of the Workplace Safety and Insurance Board; or

- as the result of an accident incurred in the performance of duty; or
- while the employee is absent on sick leave due to illness if the illness occurs before the employee goes off duty to commence their scheduled vacation; or
- Employees who become hospitalized during an approved vacation, may substitute such sick leave as they have owing to them and will be granted alternative vacation days equivalent to the number of vacation days hospitalized (excluding non-scheduled work days) providing that:
 - i) they were hospitalized in a recognized institution and verification of this is received by Management.
 - ii) the alternative days are taken at a time mutually convenient to the employee and Management.

In all other cases, the scheduled vacation will be deemed to have been taken as scheduled, and no substitution of sick leave for vacation shall be permitted.

17.2 Vacation – Part-Time Employees Only

Employees covered by this Collective Agreement shall receive vacation with pay as follows:

- Up to one year of service, one day vacation for each month of service with pay at the rate of 4% of their hourly rate of pay for all hours worked.
- One year of service, three weeks' vacation with pay at the rate of 6% of their hourly rate of pay for all hours worked.
- An employee with seven (7) years of service, four weeks' vacation with pay at the rate of 8% of their hourly rate of pay for all hours worked.
- An employee who has at least 15 years of continuous past service as at March 31 in any year shall be entitled to five (5) weeks of vacation with pay at the rate of 10% of their hourly rate of pay for all hours worked.
- An employee with twenty-five (25) years of service, six weeks of vacation with pay at the rate of 12% of their hourly rate of pay for all hours worked
- It is understood that the service year shall be as at March 31st.

ARTICLE 18 - COMPASSIONATE LEAVE AND LEAVE OF ABSENCE

18.0 The Employer may grant leave of absence without pay but continued benefits as applicable upon request by the employee for a period of up to 12 days, provided that such absence may be accommodated within the work schedule without additional cost. Employees will make requests for such absences at least 7 calendar days in advance, except in the case of legitimate personal emergencies. Such leave may be extended upon request by the employee at the Employer's sole discretion. Employees who are absent resulting from such leave of absence shall not be considered to be laid off, and their seniority shall continue to accumulate during such absence.

18.1 Bereavement Leave

In the event of the death of an employee's wife, husband, partner, child, grandchild, parent, sister or brother, the employer, at the request of the employee will arrange leave of absence with pay, such period not to exceed five days, excluding scheduled days off. Bereavement leave days taken by part time employees must be taken within a fourteen day calendar period.

In the event of the death of an employee's mother-in-law, father-in-law, brother-in-law, sister-in-law, the Employer at the request of the employee, will arrange leave of absence with pay, such period not to exceed three (3) days concluding for Part-time employees on the day of the funeral. In the event of the death of an employee's grandmother or grandfather, the Employer at the request of the employee will arrange leave of absence with pay for a period not to exceed two (2) days concluding for Part-time employees on the day of the funeral. In the event of the death of an employee's aunt or uncle, the Employer at the request of the employee will arrange leave of absence with pay for a period not to exceed one (1) day, concluding for Part-time employees on the day of the funeral. Any such request will be in writing on forms supplied by the Employer, but because of the nature of the said leave, such forms may be filled in by the employee after returning to work.

Pay for bereavement leaves are for scheduled shifts only.

Bereavement Leaves included in this Article may be extended by two paid days under exceptional circumstances at the discretion of the Home Administrator. In addition, the Home Administrator may grant a one day paid leave for other occasions of bereavement.

All relationships listed in this article will be understood to include "step" relationships and relationships associated with an employee's common-law spouse or partner of the same sex.

In the event bereavement leave is required while an employee is on paid vacation, such paid vacation shall be reaccredited to the employee. Management reserves the right to request proof.

18.2 Pregnancy And Parental Leave

- (a) For the purposes of Article 18.2, "with pay" and "regular earnings" shall refer to the regular straight time wage or scheduled hours or, in the event of extended leaves, the average hours worked in the preceding twenty week period. For part-time employees, this rate shall include the allowance for pay-in-lieu of benefits and vacation.
 - (i) Pregnancy and parental leaves will be granted in accordance with the *Employment Standards Act, 2000* (as amended from time to time) of Ontario.
 - (ii) Pregnancy leave shall be granted for up to seventeen (17) weeks in duration (or as otherwise provided by the Employment Standards Act, 2000) and may begin no earlier than seventeen (17) weeks before the expected birth date (or as otherwise provided by the Employment Standards Act, 2000).
 - (iii) Parental leave shall be granted for up to sixty-one (61) weeks for an employee who also takes a pregnancy leave or sixty-three (63) weeks for an employee who does not take a pregnancy leave (or as otherwise provided by the Employment Standards Act, 2000) and may begin no later than seventy-eight (78) weeks after the day the child is born or comes into the employee's custody, care and control for the first time (or as otherwise provided by the Employment Standards Act, 2000).
- (b) (i) An employee who is pregnant shall be entitled, upon application, to pregnancy leave and parental leave immediately thereafter. The employee shall give the Employer at least two (2) weeks notice, in writing, of the day upon which they intend to commence their leave of absence, unless impossible, and furnish the Employer with a certificate of a legally qualified medical practitioner stating that they are pregnant and giving the estimated day upon which delivery will occur.
 - (ii) The employee must have started employment with their Employer at least thirteen (13) weeks prior to the expected date of birth.

- (iii) The employee shall give at least two (2) weeks' notice of their intention to return to work. The employee may, with the consent of the Employer shorten the duration of the leave of absence requested under this Article upon giving the Employer two (2) weeks' notice of their intention to do so, and furnishing the Employer with a certificate of a legally qualified medical practitioner stating that they are able to resume their work. Additional leave of absence may be taken under Article 18.2 (a)(iii) Parental Leave.
- (iv) An employee on pregnancy leave who is in receipt of Employment Insurance pregnancy leave benefits shall be paid a supplemental Employment Insurance benefit. That benefit will be the equivalent to the difference between seventy-five percent (75%) of their regular weekly earnings and the sum of their weekly Employment Insurance benefits and any other earnings. Such payment shall commence following the completion of the applicable Employment Insurance waiting period and upon receipt by the Employer of the employee's Employment Insurance cheque stub as proof they are in receipt of Employment Insurance Pregnancy benefits, and shall continue while the employee is in receipt of such benefits for a maximum period of seventeen (17) weeks. The employee's regular weekly earnings shall be determined by multiplying their regular hourly rate on their last day worked prior to the commencement of the leave times their normal weekly hours.
- An employee who does not apply for leave of absence under Article 18.2(a)(ii) and who is otherwise entitled to pregnancy leave thereunder, shall be entitled to and shall be granted leave of absence in accordance with Article 18.2(a) upon providing the Employer before the expiry or two (2) weeks after they ceased to work, with a certificate of a legally qualified medical practitioner stating that they were not able to perform the duties of their employment because of a medical condition arising from their pregnancy, and giving the estimated medical day upon which, in their opinion, delivery will occur or the actual date of their delivery.
- An employee who intends to resume their employment on the expiration of the leave of absence shall so advise the Employer when they request the leave of absence. Upon their return to work following such leave an employee will be reinstated in their former job, classification, and at the rate of pay the employee would be earning if they had worked through the leave if at all practicable. Upon their return to work following such leave, an employee shall also be reinstated to the shift that they would have worked had they worked through the leave. All employees who fill vacancies as a result of the above absences shall likewise be returned to their former permanent positions.

- 18.5 When the Employer has suspended or discontinued operations during the leave of absence and has not resumed operations upon the expiry thereof, the Employer shall upon resumption of operations, reinstate the employee to their employment or to alternate work in accordance with the established seniority. System or practice of the Employer in existence at the time the leave of absence began, with no loss of seniority or benefits accrued, and shall reinstate the employee in accordance with the provisions of Article 18.4.
- 18.6 Pregnancy leave is not an illness under the interpretation of this agreement, and credits on the accumulated sick leave plan and the weekly indemnity plan cannot be used.
- 18.7 Credits for service for the purpose of salary increments, vacations or any other benefit under the Collective Agreement shall continue and seniority shall accumulate during the pregnancy and parental leave.
- 18.8 The total leave of absence provided for under Article 18.2 shall be extended upon application in writing at least two (2) weeks prior to the expiry of the leave, to a total period of seventy-eight (78) weeks following the date the leave commenced.
 - (a) An employee who becomes a parent, and who has been employed for at least thirteen (13) weeks immediately preceding the date of the birth of the child, or the date the child first came into the care or custody of the employee, shall be entitled to parental leave.
 - (b) A "parent" includes: the natural mother or father of the child; a person with whom a child is placed for adoption and a person who is in a relationship of some degree of permanence with the parent of a child and who intends to treat the child as their own.
 - (c) Parental leave must begin no later than seventy-eight (78) weeks after the child is born or comes into the employee's custody, care and control of the first time (or as otherwise provided by the *Employment Standards* Act, 2000). For employees on pregnancy leave, parental leave will begin immediately after pregnancy leave expires.
 - (d) For the purposes of parental leave under Article 18.2(a)(iii) Parental Leave, the provision under 18.4, 18.5, 18.6 and 18.7 shall also apply.
 - (e) An employee on Parental Leave who is in receipt of El Parental Leave benefits shall be paid a supplemental Employment Insurance benefit. That benefit will be equivalent to the difference between seventy-five percent (75%) of the employee's regular weekly earnings and the sum of the employee's weekly Employment Insurance benefits and any other earnings. This benefit will be calculated as the difference between (i)

seventy-five percent (75%) of their regular weekly earnings and the sum of their weekly Employment Insurance benefits and any other earnings and (ii) the weekly Employment Insurance benefit that is payable or would be payable to the employee without regard to any election by the employee to receive a lower El benefit spread over a longer period of time as may be permitted under the Employment Insurance Act. Such payment shall commence following the completion of the applicable Employment Insurance waiting period and upon receipt by the Employer of the employee's Employment Insurance cheque stub as proof that the employee is in receipt of El Parental Leave benefits, and shall continue while the employee is in receipt of such benefits to a maximum of 35 weeks.

(f) Verification and/or other evidence of the adoption may be required by the employer.

18.9 Pregnancy and Parental Leave – Full – Time Employees Only

During the period of Pregnancy or Parental Leave, the Employer shall continue to pay as applicable the Employer's portion of hospital, medical, dental, weekly indemnity, group life, pension and other benefits of this agreement, provided that the employee has maintained their share of the premium payments.

18.10 Pregnancy and Parental Leave – Part – Time Employees Only

During the period of Pregnancy or Parental Leave, the Employer shall continue to pay as applicable the Employer's portion of hospital, medical, dental, weekly indemnity, group life, pension and other benefit premiums pertaining to part time staff of this agreement, provided that the employee has maintained their share of the premium payments. In the case of Part-time employees, the payment-in-lieu shall be accounted for separately under the supplemental Employment Insurance benefits.

18.11 *Jury Duty*

Where an employee, who would otherwise be on normal duty with the Corporation, is absent by reason of a summons to serve as a juror, or of a subpoena as a witness, they may at their option:

- (i) take leave of absence without pay and retain both the fee and the expense allowance paid by the Court; or
- (ii) charge the absence against vacation credits and retain both-the fee and expense allowance paid by the Court; or

(iii) treat the absence as a leave with pay (but with no charge against any of their credits) but surrender the fee to the City Treasurer when received.

Such leaves will be for a minimum of four hours, if scheduled.

18.12 Union Leave

An employee who is elected or appointed to office in the Local or National Union shall, upon request, be granted a leave of absence, with seniority rights unaffected for all purposes, for the employee's term of office, up to a maximum of three (3) years. Only one (1) employee from each Bargaining Unit may be on such leave at one time.

All such leaves are without pay or provision of benefits including vacation entitlement by the Employer, except that and conditional upon reimbursement by the Union for wage and benefit premium costs incurred, employees on such leaves shall be continued in benefit plans other than Sick Leave, Short Term or Long Term Disability benefits, or OMERS Pension benefits, and paid through the Employer's payroll system. It is further understood that, for purposes of Worker's Compensation, the Employer of record during such absences is the Union. The Union will provide a minimum of four weeks advanced notification of the return of employees from such leaves.

Leaves of a lesser period during the two year term may be granted at the discretion of the Director, People Services or designate. The Union shall give at least seven days advance notice.

18.13 Union Education Leave

The Employer shall grant leave of absence without loss of seniority or service to employees to attend Union Conventions, Seminars, Educational Classes, or other Union business. In making application for leave of absence for Union business, it is understood that a leave of absence shall be for no longer than a two (2) week period in one (1) calendar year. Where leave of absence for Union business is requested, it is understood that the Union will not request leave of absence for more than four (4) employees from each bargaining unit at any one time. Nor will the request be more than four (4) employees from Nursing and no more than two (2) employees from each other section at any one time. The total leave period in one year will not exceed fifty (50) scheduled shifts in total, excluding leave time for the Chairperson.

The Union shall be responsible for the payment of wages during the time of absence. Upon receipt of an invoice from the Employer the Union shall reimburse the Employer for wage payments advanced to employees on Union leave.

18.14 Education Leave

Employees may request leaves of absence for purposes of educational programs, to a maximum of two years. Such requests shall be submitted in writing to the Home Administrator (or designate), stating the period of requested absence and program details, and be subject to the approval of the Home Administrator (or designate), such approval not to be unreasonably withheld. Employees must submit proof of registration in the educational program. Such leaves shall be unpaid and without benefit continuation, however seniority will be frozen and retained during the approved absence.

The approved leave of absence will continue only as long as the employee remains enrolled in the program.

18.15 General

Employees who are on leave of absence will not engage in gainful employment while on such leave and if an employee does engage in gainful employment while on such leave of absence, they may forfeit all seniority rights and privileges contained in this Agreement.

ARTICLE 19 - HOSPITAL, MEDICAL, PENSION, INSURANCE PROGRAM

19.0 Entitlement to Hospital, Medical, Pension and Insurance programs will be as more particularly described in Article 19.1 for Full – Time employees and Article 19.2 for Part-Time employees.

19.1 Full – Time Employees Only

- (a) The Employer will pay 100% of the premiums for the following benefit programs:
 - The Liberty Health Supplementary Plan.
 - The Liberty Health Extended Health Care Benefits Plan, including the Deluxe Travel Plan.
 - The maximum prescription dispensing fee covered will be \$9.00. Any drug plan will provide for generic substituting except upon express instruction of a physician.

- The Liberty Health Extended Health Care Benefit Program will include a rider providing a benefit for the provision of vision care and hearing aids.
- The Liberty Heath Extended Health Care Benefit Program rider which
 provides vision care and hearing aid benefit will provide a two hundred
 and eighty five dollar (\$285.00) vision care benefit every twenty-four
 (24) months, with a twelve (12) month prescription and loss or breakage
 replacement up to the maximum dollar amount, and a three hundred
 dollar (\$300.00) once in every five years hearing aid benefit. The vision
 care benefit can be used toward the cost of corrective eye surgery.
- An eye exam once every twenty-four (24) consecutive months up to a maximum amount of eighty dollars (\$80.00) per exam for employees and covered dependents only.
- Registered Massage therapist maximum amount allowable two hundred dollars (\$200.00) per insured person per benefit year and Chiropractic services – maximum amount allowable two hundred dollars (\$200.00) per insured person per benefit year. Effective November 14, 2019, the above registered massage therapist and chiropractic benefit ceases and the Employer to provide a combined paramedical benefit of \$700.00 (effective July 1, 2020 \$1000.00) per person in any 12 consecutive months for speech pathologist, physiotherapy, chiropractor and registered massage therapy.

Details and conditions of the Extended Health Care and the Deluxe Travel Plans are set out in the insurance contract and the benefit brochures.

Enrolment in the Liberty Health benefit plans will be effective the date of hire of the employee.

- (b) The Employer agrees to contribute to the Ontario Municipal Employees Retirement System; enrolment in the Plan to be optional for employees taken on staff prior to May 1, 1964, and mandatory for employees employed after May 1, 1964, as established under the relevant municipal by-law.
- (c) The Employer agrees to pay 100% of the premium of the Group Life Insurance Plan for those employees who wish to participate. The amount of life insurance shall be 1.5 times annual salary to a maximum of \$75,000 for all employees. The Disability Waiver of premium benefit shall be provided to age 65.
- (d) The Employer agrees to pay 80% of the premium cost of the Blue Cross No. 9 Dental Care Plan at constantly maintained current O.D.A. fee schedule benefits, and Dental Rider #3 (Orthodontic) on a 50/50 coinsurance basis with a \$2,000.00 lifetime maximum.

- (e) Every employee shall advise the Employer of their current marital and dependent status for the purposes of benefit entitlement and will thereafter be responsible to keep the Employer informed of changes in that status as such changes may arise.
- (f) It is understood and agreed by the Parties that the life insurance referred to in this Article requires a 90 day waiting period from the time of commencing employment until the insurance is placed in force. No claim may be presented or allowed until after the expiration of this waiting period.
- (g) The Corporation has the exclusive right to select the insurer(s) provided the insurer(s) maintain equal or better benefit coverage.
- (h) Permanent active employees and new permanent employees upon completion of probationary period will receive an income indemnity insurance (short term disability) under which insurance an employee who is sick and unable to work will be paid 60% of gross pay or the weekly El maximum, whichever is greater, at the time of disability. Benefits will commence following a seven (7) day waiting period or upon exhaustion of sick credits whichever is later and shall be paid for a maximum of 26 weeks. The Corporation will pay 100% of the premium and sick payment which will constitute the Employers obligation respecting the 5/12 portion of the UIC premium reduction rebate owing to employees. Effective June 1, 2002 Short Term Disability benefit will increase to 66 2/3% of gross pay at the time of disability.
- (i) All permanent active employees and new permanent employees upon completion of the probationary period will receive a Long Term Disability Insurance Plan under which insurance an employee who is sick and unable to work will be paid 65% of gross pay at the time of disability. Benefits under the long term disability plan shall commence 196 days after the employee becomes disabled, or upon the exhaustion of sick credits together with any weekly indemnity benefits whichever is later. The benefit shall continue until the earlier of recovery, death, OMERS 90 factor eligibility or age 65. Further information in respect of this plan is set out in Schedule B of this agreement. The Corporation will pay 100% of the premium cost of the Long Term Disability insurance. Effective June 1, 2002 the Long Term Disability Insurance Benefit will increase to 70% of gross pay at the time of disability with a monthly maximum of \$2500.
- (j) For clarity, full-time employees over the age of 65 are entitled to the benefits in Article 19.1 with the exception of the benefits in article 19.1(c) (Group Life Insurance Plan) and 19.1(i) (Long Term Disability Insurance Plan).

(k) For new retirees:

The employer will provide for and pay the premium for a Retirees Extended Health Care Plan including the Vision care and Deluxe Travel Plan in Article 19.1(a). This benefit is available only to employees who were hired on or before August 3, 2008 and who retire after March 1, 1993 with a minimum of 20 years of service with the employer and who are at least 55 years old at the time of retirement. This Retirees Extended health Care benefit ceases when the retired employee attains age 65 years.

19.2 Part – Time Employees Only

(a) The employees covered by this Collective Agreement shall be paid thirteen and one half per cent (13.5%) (effective November 14, 2019, fourteen per cent (14%); effective July 1, 2020, fourteen and one half percent (14.5%)) of their hourly rate for each hour worked, payable bi-weekly, in lieu of not receiving benefits payable to Full-time Service Employees, which benefits include - Sick Leave - Semi-Private Hospital Care - Extended Health Care - Dental Care - Life Insurance - and any other benefits not specifically mentioned in this Collective Agreement but set out in the Collective Agreement for Full-Time Service Employees.

Part time employees replacing full-time employees on a temporary basis under Article 22 will continue to be paid the percentage in lieu of receiving benefits payable to permanent full-time employees.

- (b) Full-time employees who transfer to part-time status shall remain members of the Ontario Municipal Employees Retirement System. (OMERS)
 - Effective May 25, 1992, employer contributions towards OMERS plans for employees who qualify for, and participate in the pension plan shall not be deducted from pay in lieu of benefits under 19.2(a).
- (c) Chiropractor's certificates supporting an employee's inability to attend work will be accepted by the Home management if recognized by competent authorities such as the Ontario Ministry of Health.

ARTICLE 20 - REPORTING PROCEDURES

20.0 Reporting procedures for each section will be established by the Home Administrator. It is the responsibility of the employee to notify their immediate Management Supervisor in respect to time off other than regular assigned days

off, and in no case will time be paid for where the established reporting procedures are not followed or reason satisfactory to the Employer provided.

ARTICLE 21 - POSTING OF STAFF VACANCY

21.1 (a) It is mutually agreed that notices of vacancies relating to jobs within the scope of the respective Bargaining Unit, shall be posted (within10 days) of the previous employee's notice of termination being submitted or immediately, when a new position is created on the Bulletin Board for a period of ten (10) days. It is understood that Paid Holidays will not be included in the aforementioned time limits. Notice of vacancy shall include qualifications for the job and wages paid. The Employer shall notify the Union Chairperson or designate on a weekly basis of all appointments involving current Union members making application for posted vacancies within the scope of the respective Bargaining Unit. Such weekly notice will also be posted in the workplace.

Where vacancies are posted for positions within the Full-time and Part-time units, consideration will be given to applications from both Full-time and Part-time employees to fill such vacancies. The seniority of applicants will be observed for such purposes provided the senior applicant possesses the necessary qualifications and ability to perform the work available. If there is no successful applicant within the bargaining units, the Home may consider persons outside the bargaining units.

The Employer shall provide the Union Chairperson or designate with all job and line postings.

- 21.2 Where employees transfer from the Full-time Bargaining Unit to the Part-time Bargaining Unit, without break in service, the following terms and conditions apply.
 - (a) Seniority credits earned within the Full-time Unit shall be transferred to the Part-time Unit, and shall thereafter be expressed in hours calculated by multiplying the number of completed months of service in the Full-time Unit by 173, and adding to that number the hours actually worked in any remaining period of service.
 - (b) Employees who transfer without changing classification shall not be subject to the trial period specified under Article 12.8, but progression through the salary range provided for the classification would be based on seniority as determined by part (a) of this Sub-Article.

Employees who transfer, and change classifications, shall be subject to the trial period and wage progressions provided in Schedule "A".

- (c) Vacation credits earned by the employee, and not used prior to transfer, shall be paid to the employee.
- (d) Paid holiday credits earned by the employee, and not used prior to transfer, shall be paid to the employee.
- (e) Employees who, at the time of transfer were employed in a Nursing Classification, will be paid any uniform allowance earned to the time of transfer.
- (f) Sick leave credits earned to the time of transfer shall be frozen.

While the employee remains in the Part-time Unit, such credits shall have no application.

If the employee returns to the Full-time Unit, such credits shall be restored.

If the employee terminates employment, or dies, entitlement to any sick leave gratuity or pay out shall be determined solely on the basis of service within the Full-time Unit. The foregoing applies regardless if the employee was a member of the Full-time or Part-time Unit at the time of termination or death. Entitlement shall be calculated based on earnings at the time of separation.

Under no circumstances is an employee entitled to any payment of sick leave credits for reason only that they transferred to the Part-time Bargaining Unit.

- (g) Full-time employees who transfer to part time status shall remain members of the Ontario Municipal Employees Retirement System.
- 21.3 (a) Part-time employees who become Full-time employees, regardless of job classification and without interruption of continuous service, shall receive seniority credit, for all purposes other than job postings, based on their continuous Part-time service on a pro-rata basis to the nearest even full month. To calculate a seniority date for the Part-time employee, the number of working days (full shifts) worked in the same classification shall be totaled and divided by twenty, and taken to the higher total, if not an exact amount.

For posting purposes only, all part time and full time employees, regardless of job classification and without interruption of continuous service, shall receive seniority credit based on their date of hire.

- (b) When such transfer takes place on any day during the year, other than April 1, the provisions of 17.1 (e) – vacation - of the Full-time agreement shall apply until March 31 next following in lieu of the fact that the Parttime employee has been compensated for vacation earned in their biweekly pay cheque pursuant to Article 17.2 of the Part-time agreement. Thereafter, the regular provisions of article 17.1 of the Full-time Agreement shall apply.
- (c) Similarly, when any such transfer takes place, all of the provisions of Articles 16.3 and 19.1 of the Full-time unit agreement shall come into effect for such employees on the first day of the month following such transfer and no retroactivity of benefits shall apply.
- (d) Notwithstanding the right of transfer of seniority, the trial period provisions shall continue to apply.
- (e) To calculate seniority days for a Full-time employee who becomes a Parttime employee without interruption of continuous service, two hundred and sixty one days will be given for each year of continuous service, plus twenty one days for each additional month, to the nearest full month.
- (f) A day (i.e., a full shift) is considered to be eight hours. Shifts of lesser duration would be considered to be a portion of a day, and would be cumulative.

<u>ARTICLE 22 - FILLING OF TEMPORARY FULL-TIME POSITIONS - PART-TIME</u> EMPLOYEES ONLY

Opportunities will be made available first to regular Part-time employees and second to casual part time employees to replace Full-time employees, when the latter are on long term temporary leaves of absence (any absence eight (8) weeks or greater), such as maternity, parental, WSIB, education leave, or sick leave and are expected to return to their former position. In accordance with the following; such opportunities will be limited to twenty-six (26) weeks per Part-time employee per opportunity.

For the purpose of clarity, the only person to move to the bottom of the list will be the person who has accepted the opportunity.

New staff or staff transferring into the classification will be added to the bottom of the list.

- (a) Notices will be placed on a notice board outlining the opportunities to replace Full-time employees who are on long term absence, by job and in chronological order by start date of the opportunity.
- (b) Regular Part-time employees currently working in the same job class as that posted may apply to be considered for the opportunity within five (5) calendar days of the posting. Part-time employees who are currently on a Full-time assignment are ineligible to apply.
- (c) Such opportunities shall be offered on the basis of seniority, except that when a senior employee has accepted an opportunity, they will thereafter be placed at the bottom of a constantly maintained list. When notice of more than one such opportunity is posted, the employee who is top-most on the constantly maintained list will be offered the first chronological opportunity, the second-most the second opportunity and so on.
- (d) Should no regular part time employee accept an opportunity, the opportunity will be provided to casual part time employees utilizing the same process outlined in (b) and (c) above.

<u>22.1 - FILLING OF TEMPORARY REGULAR PART-TIME POSITIONS - CASUAL PART-TIME EMPLOYEES ONLY</u>

Opportunities will be made available to Casual Part-time employees to replace Regular Part-time employees, when the latter are on long term temporary leaves of absence (any absence twelve (12) weeks or greater), such as maternity, parental, WSIB, education leave, sick leave or are covering a long term full time vacancy and are expected to return to their former position. In accordance with the following; such opportunities will be limited to twenty six (26) weeks per Casual Part-time employee per opportunity.

For the purpose of clarity, the only person to move to the bottom of the list will be the person who has accepted the opportunity.

New staff or staff transferring into the classification will be added to the bottom of the list.

(a) Notices will be placed on a notice board outlining the opportunities to replace regular part-time employees who are on long term absence, by job and in chronological order by start date of the opportunity.

- (b) Casual Part-time employees currently working in the same job class as that posted may apply to be considered for the opportunity within five (5) calendar days of the posting. Casual Part-time employees who are currently on a full-time or regular part time assignment are ineligible to apply.
- (c) Such opportunities shall be offered on the basis of seniority, except that when a senior employee has accepted an opportunity, they will thereafter be placed at the bottom of a constantly maintained list. When notice of more than one such opportunity is posted, the employee who is top-most on the constantly maintained list will be offered the first chronological opportunity, the second-most the second opportunity and so on.

ARTICLE 23 - UNIFORMS

- 23.0 It is understood and agreed that where employees, other than employees in nursing classifications, are required to wear uniforms, the Employer will supply 4 uniforms to full-time employees and 3 uniforms to part-time employees on an annual basis. The Employer will supply one pair of safety shoes to personnel designated by the Employer, and will replace such shoes as required.
- 23.1 The Employer will supply 4 uniforms annually for staff in a full time capacity in a nursing classification. The Employer will supply two (2) uniforms for staff in a part time capacity in a nursing classification.

ARTICLE 24 - MEDICAL EXAMINATION

- 24.0 (a) The Employer shall pay the cost of an employee's initial compulsory employment medical examination by a physician designated by the Employer.
 - (b) If the Employer has reasonable cause to be concerned about an employee's health, it may require the employee to undergo a medical examination by a physician of the Employer's choice at the Employer's expense.
- 24.1 Costs of a required annual medical examination will be either:
 - (a) 75% paid by the Employer and 25% paid by the employee (if the employee chooses their own physician) to a maximum cost of \$50.00 paid by the employer.
 - (b) Paid by the Employer (if medical by Employer's physician).

ARTICLE 25- TERMINATION OF EMPLOYMENT

25.0 The Employment Standards Act will apply to cases of dismissal, lay-off and termination of employees.

ARTICLE 26 - TRAINING AND EDUCATIONAL COURSES

- 26.0 Where the Employer requires an employee to attend a training program, the Employer shall pay the tuition cost of such program, and the time spent in attendance shall be considered time worked.
- 26.1 The Corporation will reimburse an employee one hundred percent (100%) of the tuition cost of a course of instruction taken by such employee to better qualify themselves to perform their job, provided the approval of the Home Administrator (or designate) is obtained prior to the commencement of such course and the employee successfully completes the course. Reimbursement will be made when the employee presents proof of successful completion.
- 26.2 Professional College of Nurses registration fees for RPN's will be paid as follows:
 - Full time 100%
 - Part time 50% or the pro-rated amount based on the actual hours worked in the previous year as a percentage of full time hours (2080 hours/year), whichever is greater.

ARTICLE 27 - LETTERS OF REPRIMAND

27.0 All letters of reprimand will be removed from an employee's file after eighteen (18) months providing that there have been no similar infractions during that eighteen (18) month period.

ARTICLE 28 - ADVICE TO THE EMPLOYER

28.0 Each Employee has a responsibility to provide to the Employer the address of the employee's personal residence and the number of the employee's telephone at such residence, and to immediately advise the employer of any change to either.

ARTICLE 29 - PERSONNEL FILES

29.0 An employee shall, upon written request made a reasonable amount of time before viewing, have an opportunity to view their personnel file in the presence of the Director, People Services or designate. If so desired by the employee, they may be accompanied by a representative of the Bargaining Unit. The information which the employee may view will be:

- a) written application form
- b) written evaluations
- c) formal disciplinary letters
- d) performance appraisals.

ARTICLE 30 - RETROACTIVE PAY

Any wage increases shall be effective as and from the dates specifically listed on a retroactive basis to all employees, covered by the collective agreement, for all paid hours of employment. Any new employees shall be entitled to a pro rata adjustment to their remuneration from the date of their employment. The Home shall be responsible to contact, in writing, with a copy to the Union, at their last known address, employees who have left its employ, to advise them of their entitlement to any retroactive wage adjustment. Any employee who has terminated employment prior to the signing of this Agreement shall have a period of 60 days only from the date of the execution of the collective Agreement in which to claim from the Home any adjustment to remuneration. The retroactive payments shall be made to all employees so entitled within 60 days.

ARTICLE 31 - MILEAGE ALLOWANCE

- 31.0 When a member of the Bargaining Unit is required to use their car for the Employer's business, they shall be paid in accordance with the prevailing City of London policies concerning mileage reimbursement.
- 31.1 Should the mileage allowance spelled out in paragraph (a) of this Article be increased in accordance with the City of London Policies, then it will apply to the employees at Dearness Home.

ARTICLE 32 – RPN ALLOWANCE

Where an Registered Nurse is absent from their normal shift, and the Employer temporarily assigns an Registered Practical Nurse to carry out some additional responsibilities of the absent Registered Nurse for a period in excess of a half (1/2) shift, the employee shall receive an allowance of ten dollars (\$10.00) for each shift from the time of assignment. In any event, a Registered Practical Nurse will not

be temporarily assigned to carry out such additional responsibilities on the night shift.

ARTICLE 33 - TERM OF AGREEMENT

- 33.0 This Agreement shall be for a three (3) year term, commencing on the 1st day of July, 2019 and ending on the 30th day of June, 2022 and thereafter in each succeeding year, subject to changes and amendments agreed to by both Parties in writing.
- 33.1 Either party to this Agreement may, within the period of ninety (90) days before the termination of the Agreement, give notice in writing to the other Party of its desire to bargain with a view to the renewal, with or without modifications, of this Agreement. Within such period as the Parties agree upon following the giving of such notice, they shall bargain in good faith and make every reasonable effort to arrive at a Collective Agreement.

IN WITNESS OF the Parties have executed this Agreement:

CITY SOLICITOR'S
OFFICE
CITY OF LONDON

DATE: Only 29/12

APPROVED AS TO FORM ONLY

FOR THE CORPORATION OF THE CITY OF LONDON:

Mayor	Jesse Helmer, Deputy Mayor	
City Clerk	Catharine Saunders, City Clerk	

SEP 0 3 2020

Date
SEP 0 3 2020

Date

FOR UNIFOR, LOCAL 302

May Rymal

Marte Su A

Gay Care

2 Bayers

Date

SCHEDULE "A" FULL-TIME AND PART-TIME SERVICE WORKERS WAGES

Position	Effective Date	Start Rate	1040 Hrs 6 Month Rate	2080 Hrs 1 Year Rate
R.P.N. (DS011)	July 1, 2019	27.88	28.24	29.35
R.P.N Wound Care (DS012)	July 1, 2020	28.16	28.52	29.64
R.P.N MDS/RAI Coordinator (DS013)	July 1, 2021	28.44	28.81	29.94
Recreation Coordinator (DS151)	July 1, 2019	27.44	27.76	28.88
Volunteer Coordinator (DS152)	July 1, 2020	27.71	28.04	29.17
	July 1, 2021	27.99	28.32	29.46
	July 1, 2019	23.06	23.28	24.09
Resident Assistant (DS141)	July 1, 2020	23.29	23.51	24.33
	July 1, 2021	23.52	23.75	24.57
Dietary Aide (DS041)	July 1, 2019	22.46	22.78	23.49
Laundry Aide (DS042)	July 1, 2020	22.68	23.01	23.72
	July 1, 2021	22.91	23.24	23.96
10.00	July 1, 2019	22.51		23.49
Cafeteria Food Serv Gen (DS043)	July 1, 2020	22.74		23.72
	July 1, 2021	22.97		23.96
Kitchen Help (DS052)	July 1, 2019	22.51	22.79	23.49
Housekeeping Aide (DS051)	July 1, 2020	22.74	23.02	23.72
	July 1, 2021	22.97	23.25	23.96
	July 1, 2019	25.12	25.38	26.24
Cooks (DS061)	July 1, 2020	25.37	25.63	26.50
	July 1, 2021	25.62	25.89	26.77
Dana and Commant Wards at U.S.	July 1, 2019	22.92	23.20	24.25
Personal Support Worker/Health Care	July 1, 2020	23.25	23.53	24.59
Aide (DS032)*	July 1, 2021	23.48	23.77	24.84

SCHEDULE "A" (continued) FULL-TIME AND PART-TIME SERVICE WORKERS WAGES

Position	Effective Date	Start Rate	1040 Hrs 6 Month Rate	2080 Hrs 1 Year Rate
	July 1, 2019	23.06	23.28	24.32
Groundskeeper (Seasonal) (DS091)	July 1, 2020	23.29	23.51	24.56
20 20 20 40	July 1, 2021	23.52	23.75	24.81
Laundry Person (DS081)	July 1, 2019	22.91	23.18	24.06
Shipper/Receiver (DS082)	July 1, 2020	23.14	23.41	24.30
, , , , , , , , , , , , , , , , , , , ,	July 1, 2021	23.37	23.64	24.54
	July 1, 2019	24.94	25.25	26.24
Maintenance Person (DS101)	July 1, 2020	25.19	25.50	26.50
	July 1, 2021	25.44	25.76	26.77
	July 1, 2019	27.05	27.33	28.39
Carpenter (DS111)	July 1, 2020	27.32	27.60	28.67
	July 1, 2021	27.59	27.88	28.96
	July 1, 2019	28.05	28.40	29.41
Electrician (DS121)	July 1, 2020	28.33	28.68	29.70
	July 1, 2021	28.61	28.97	30.00
	July 1, 2019	27.09		28.30
Groundskeeper (DS132)	July 1, 2020	27.36		28.58
	July 1, 2021	27.63		28.87

*NOTE:

The Personal Support Worker/Health Care Aide classification reflects both Personal Support Workers and Health Care Aides. This classification title amendment does not grant employees within the former Health Care Aide classification who are not certified as a Personal Support Worker as being deemed by the Corporation to be qualified/certified as a Personal Support Worker.

Schedule A has been adjusted to reflect the following one-time adjustments for the Personal Support Worker/Health Care Aide:

- Effective July 1, 2019 a one-time adjustment of \$0.15/hour
- Effective July 1, 2020 a one-time adjustment of \$0.10/hour

One-time Lump Sum Payments

- 1. Within the period of October 16, 2019 to December 14, 2019, a one-time lump sum payment of four hundred dollars (\$400.00) less all applicable statutory deductions required by law (prorated for part-time employees and amount is not subject to percentage in lieu of receiving benefits or percentage in lieu of vacation pay as provided for under Article 19.2) shall be paid by the Corporation to all Unifor Local 302 bargaining unit members who are employed by the Corporation, and actively at work with the Corporation as at date of ratification. For employees not actively at work, these employees will receive the lump sum payment upon return to work provided that the return to work occurs within the term of the Collective Agreement and in any event is no later than June 30, 2022.
- 2. Effective July 1, 2020 a one-time lump sum payment of three hundred dollars (\$300.00) less all applicable statutory deductions required by law (prorated for part-time employees and amount is not subject to percentage in lieu of receiving benefits or percentage in lieu of vacation pay as provided for under Article 19.2) shall be paid by the Corporation to all Unifor Local 302 bargaining unit members who are employed by the Corporation, and actively at work with the Corporation on July 1, 2020. For employees not actively at work, these employees will receive the lump sum payment upon return to work provided that the return to work occurs within the term of the Collective Agreement and in any event is no later than June 30, 2022.

SCHEDULE "B" LONG TERM DISABILITY DETAILS

ELIGIBILITY

All full time active employees. New employees upon

completion of probation.

COMPULSORY BENEFIT Yes

WAITING PERIOD

Minimum of 196 calendar days but not before exhaustion of

sick leave and weekly indemnity benefits.

BENEFIT LEVEL

70% of salary at the time of disability. (Effective June 1, 2002)

MAXIMUM BENEFIT

\$2500 per month (Effective June 1, 2002)

BENEFIT DURATION

The earlier of recovery, death, OMERS, 90 factor or age 65.

OFFSETS

Employee Canada Pension Benefit (not dependant) and Workplace Safety and Insurance if 70% benefit level. If 85% - all income sources participated in by the employer and employee, eg. OMERS disability, Canada Pension disability benefit and Workplace Safety and Insurance benefits if

applicable.

PRE- EXISTING CONDITIONS

This clause is waived.

OWN OCCUPATION

2 years own occupation. Beyond that, any occupation. Also

provides for a two year rehabilitation period with a 50%

benefit level.

RESTRICTIONS

NERVOUS AND MENTAL Must be under the care of a psychiatrist - if not, benefit could

be reduced to a 50% level.

WAIVER OF PREMIUM

No premium is payable if eligible for benefit.

UNDERWRITING

METHOD

Fully pooled as opposed to experience rating.

PREMIUM SHARING

100% paid by the employer.

IMPLEMENTATION DATE January 1, 1998 with revisions June 1, 2002.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (Dearness Services)

- and -

UNIFOR, LOCAL NO. 302

The following Letters of Understanding have been renewed and re-dated by the Parties and compiled into this document as follows:

Index Letter No.	Subject
2019-01	Format of Collective Agreement
2019-02	Call-ins and Overtime
2019-03	Temporary Full Time Hours
2019-04	Emergency Call-in List
2019-05	Scheduling Times and Rotation
2019-06	Education Leave
2019-07	Regular Part-Time and Casual Part-Time
2019-08	Resident Abuse
2019-09	Women's Advocate
2019-10	Outbreaks in the Workplace
2019-11	Diversity and Inclusion
2019-12	Uniforms
2019-13	Summer Students - PSW/HCA

All letters of understanding listed append to and become part of the Collective Agreement.

For the Employer	For the Union
	May Rymal
	any cot
	the leb
	58

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Format of the Collective Agreement

In order to provide for better administration of the Collective Agreement and to facilitate future negotiations concerning the renewal of the Agreements, the Parties have agreed to publish combined part time and full time Collective Agreements.

The parties under recognize that this format is being utilized as a convenience only, and in no way changes the pre-existing rights of either Party under the separate Collective Agreements. The change in format per se will not be relied on by either Party, in the event of future grievances or arbitration, in interpreting the provisions of the Collective Agreement.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Letter of Understanding – Call-in and Overtime

Call-In Guidelines

- 1. When calling in part time staff, call phone number listed by the staff person's name.
- 2. Names are listed in order of seniority. Shift preferences as listed; will try to acknowledge this but when short, call for any shift.
- 3. Call-in list will be updated twice a year (June 30th and December 31st).
- A five (5) minute wait time will be used between call-ins to allow staff time to respond with the exception of 21:00 to 7:00 Monday to Friday and 18:00 to 07:00 Saturday and Sunday.
- 5. If the shift has to be filled within a three (3) hour period of the start of the shift, the 5 minute wait time will not be required.
- 6. Call-ins received will be offered to all eligible part time employees in order of seniority using a rotating seniority list. Work opportunities shall be offered as received. The most senior part time employee will be offered a choice and if more than one work opportunity is required, the part time employee will only be able to accept one shift, excluding Saturdays and Sundays.
- 7. Overtime offered to staff on duty to cover a shift that was not able to be covered through the call-in procedure must be done by seniority. The scheduler or manager/supervisor will offer overtime to the most senior part time staff working and will continue down the seniority list until the shift is accepted. If no part time staff accepts, the offer will go to the most senior full time staff working and will continue down the seniority list until the shift has been accepted.

- 8. Any demonstrated inequity regarding overtime opportunities brought to the attention of management will be remedied. The remedy will be that the next available overtime opportunity will be offered to the individual. This offer will be extended up to three (3) times.
- 9. Employees on the call-in list, who do not accept a call-in in a three (3) month period shall be removed from the call-in list for the following six (6) months.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Temporary Full-Time Hours

Notwithstanding the criteria of Article 22 of the Collective Agreement, opportunities for temporary full-time hours during the summer months (mid June to mid September) will be made available for the following positions: 15 PSW/HCA, 6 RPNs, 2 Dietary Aides, 2 Housekeeping Aides, 2 Laundry Aides and 1 Cook.

These vacancies will be filled by using the rotating seniority list as outlined in Article 22. These opportunities will be made available to cover full-time staff while they are on vacation over the summer months.

Successful incumbents will only be eligible for one week (5 days) of vacation during the summer months in which they are working in the temporary full time position.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES) - and -

UNIFOR, LOCAL 302

Letter of Understanding - Emergency Call-In List

- Part time worker may be placed on an "Emergency Call-in List" for classifications other than their own provided that they are qualified to perform the duties with minimal training and orientation.
- 2. The following classifications will be limited to the specified numbers of workers to be placed on the list:

Housekeeping 7 workers Laundry 5 workers Activities 3 workers Cooks 2 workers Dietary Aide 7 workers Kitchen Help 5 workers Personal Support Worker/Health Care Aide 7 workers Maintenance 2 workers

- 3. A part time worker may be placed on no more than two (2) lists, and cannot be placed on the list for their own classification.
- 4. These lists can only be utilized after the Call-in procedure and the Overtime procedure as specified in the collective agreement have been exhausted.
- 5. The most senior part time applicant(s) will be awarded the position on the list. If a worker on the emergency list declines 3 consecutive emergency shifts, they will be removed from the list. The vacant position on the list will be posted for 14 days. The most senior part time worker will then be awarded the vacant position.
- 6. The rate of pay will be the top rate for the emergency classification.
- 7. The worker's home classification will have priority for call-ins.
- 8. Under no circumstances will a worker be awarded a permanent position in their emergency area due to any extra experience gained through this Letter. All permanent postings will continue as per the collective agreement.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

<u>Letter of Understanding – Scheduling Times and Rotation</u>

Regular Part Time Employees Only

- Regular Part Time employees are those persons who are regularly scheduled at least 32 hours and up to 40 hours in a two week period and who have committed to a part time scheduled line.
- 2. It is understood that the part time scheduled lines and/or hours of work are not attached to any posted vacancy.
- 3. All vacated part time lines which the Corporation intends to fill, or newly created lines will be posted internally within each classification as they become available as follows:
 - i. Scheduled lines will be offered to regular part time employees in each classification and granted in accordance with classification seniority,
 - Should no regular part time employee choose to accept the scheduled part time line, the vacant position will be posted in accordance with the Collective Agreement.
- 4. All lines temporarily vacated for twelve (12) weeks or more will be filled in accordance with Article 22.1.

Full Time Employees Only

1. It is understood that full time scheduled lines and/or hours of work are not attached to any posted vacancy.

- 2. All vacated full time lines, which the Corporation intends to fill, or newly created lines will be posted internally within each classification as they become available as follows:
 - Scheduled lines will be offered to full time employees in each classification and granted in accordance with classification seniority.
 - ii. Should no full time employee chose to accept the scheduled full time line, the vacant position will be posted in accordance with the Collective Agreement,
- It is understood that in order for a full time employee to be eligible to apply for a vacated line as outlined above, they must work a shift or hours of work that differ from the vacant line

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Letter of Understanding – Education Leave

If the Employer requires that an employee take courses to upgrade or acquire new employment qualifications, the employee will be entitled to paid leave of absence with no loss of seniority or benefits. In no case will an employee be paid overtime. The Employer will pay the full cost of those courses.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Regular Part-Time and Casual Part-Time

Regular Part Time employees are those persons who are regularly scheduled at least 32 hours and up to 40 hours in a two week period and who have committed to a part time line.

Casual Part Time employees are those persons who are not regularly scheduled up to 32 hours in a two week period. Article 14.10(c) of the Collective Agreement does not apply to Casual Part Time employees.

All Departments

At time of schedule preparation, if there are any available shifts beyond the part time lines, the Home will make an effort to do the following:

- first schedule these shifts for Regular Part Time employees by seniority up to a maximum of 48 hours (in a two week period);
- for those that have expressed a desire, schedule shifts by seniority up to a maximum of 64 hours (in a two week period). Employees desiring to be scheduled up to 64 hours (in a two week period) shall so advise annually by October 15th;
- 3. then, equitably distribute any shifts remaining among Casual Part Time employees by seniority.

For the purposes of scheduling as provided for in this Letter of Understanding:

- 1. Part Time employees will be deemed to be available for work as follows:
 - (a) Regular Part Time employees will be deemed to be available to be scheduled for a minimum of two (2) shift types (i.e. days, afternoons and nights). The Corporation will permit said employees to express their desire as to which two shift types they wish to be scheduled subject to operational needs and may indicate to the Scheduler in writing availability for all three (3) shift types should they wish to do so. Notwithstanding the aforesaid, Regular Part Time employees who are scheduled for the night shift will be deemed available and to have indicated a desire to be scheduled for only the night shift unless they indicate otherwise. The

- Corporation will continue to make an effort to schedule Regular Part Time employees based on their posted line.
- (b) Casual Part Time employees will be deemed available for work for all shift types (i.e. days, afternoons and nights). The Corporation will, however, make an effort to schedule Casual Part Time employees for only two (2) shift types.
- 2. Regular Part Time employees may submit a maximum of two (2) "not available" days in each two (2) week pay period within the following parameters:
 - (a) The maximum of two (2) "not available" days per pay period excludes the employees' weekend off;
 - (b) "Not available" days for the next schedule must be submitted to the Schedulers in writing prior to the applicable posted schedule preparation cut off times;
 - (c) Save and except for the maximum of two (2) "not available" days per pay period and the employees' weekend off, employees will be deemed available for work on all other days.
- 3. Casual Part time employees will be deemed available for all shifts and may be scheduled for any shift.
- 4. Regular part time employees are required to provide their "not-available" days in writing to the Schedulers prior to the applicable posted schedule preparation cut off times.
- 5. Regular part time employees that do not provide their "not-available" days will be deemed to be available for all days.
- 6. This Letter of Understanding is subject to all other provisions of the Collective Agreement except as otherwise specifically stated in this Letter of Understanding.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Resident Abuse

The parties agree that the abuse of residents will not be tolerated and that residents have a right to live in an environment that is free from abuse. For this reason, the parties agree to co-operate fully with one another in investigating any reported cases of abuse.

Where a full time employee is required to leave the workplace while an investigation is carried out in response to a complaint of abuse, such time will be with pay for all scheduled hours lost as a result of the absence.

Where a part time employee is required to leave the workplace while an investigation is carried out, such time will be with pay for all scheduled hours and reasonable missed call in opportunities as a result of the absence. The missed call in opportunity will be calculated on the basis of an averaging of the call in opportunities worked by the employee in the twenty (20) days preceding the absence.

The Employer agrees that when an employee is sent home with pay pending investigation a union committee person will be present at the time the employee is sent home provided that a union committee person is readily available. If a union committee person is not readily available, the Employer may send the employee home and advise the union of same at its earliest opportunity.

All investigations will be completed as quickly as possible. Furthermore, the parties will work to ensure there is no retaliation when an employee reports the abuse of a resident by another employee.

The parties agree that the term retaliation as used in this letter does not include counseling and/or disciplinary action taken against employees that may be issued by management as a result of an investigation.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Women's Advocate

The parties recognize that female employees may sometimes need to discuss with another woman matters such as violence or abuse at home or workplace harassment. They may also need to find out about specialized resources in the community with these and other issues.

For this reason the parties agree to recognize the role of women's advocate in the workplace. The Women's Advocate will be determined by the Union from amongst the female bargaining unit employees. The Advocate and/or Management will make themselves available to female employees as needed to discuss problems with them and access local services and supports as required.

The name of the Advocate will be posted on the union bulletin board. The Employer agrees to provide access to a private office so that confidentiality can be maintained when a female employee is meeting with the Women's Advocate.

The Women's Advocate will participate in an initial forty (40) hour training program organized by Unifor.

BETWEEN.

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Outbreaks in the Workplace

Upon recommendation of the Medical Officer of Health, all employees may take such treatments as the Officer may direct.

If an employee does not take the recommended course of treatment, or fails to complete it, they shall be placed on an unpaid leave of absence until such time as the situation is resolved. If an employee does not complete a course of treatment initiated by the Employer any subsequent course of treatment required as a result of the same situation shall be undertaken at the employee's expense.

An employee who does not take the recommended course of treatment for verified medical or bona fide religious reasons is entitled to such accommodation as the Employer may direct or, failing that sick leave or vacation if the credits are available.

Where it is permitted by the Medical Officer of Health, or designate, and where it is otherwise possible, an employee who cannot work due to not taking the recommended course of treatment may be reassigned to work in another area of the Home until the outbreak is declared over.

In the cases of employees who work at more than one health care facility, and an outbreak occurs in one of the facilities with the result being the Medical Officer of Health or designate limits the employee to working at one facility only, the Employer will attempt to offer the employee call-in hours, being respectful of the agreed to call-in procedures.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re. Diversity and Inclusion

The Corporation of City of London follows guiding principles as indicated in the Workplace Diversity and Inclusion Statement. The Corporation and the Union will respect and value the human rights and uniqueness of every individual and will continue to foster an inclusive and supportive workplace respecting the diversity, dignity, and perspectives of all.

BETWEEN.

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Uniforms

Notwithstanding Article 23.0 and Article 23.1, the parties agree that for the term of this Collective Agreement, for those employees who are required to wear uniforms, these employees shall be provided with a uniform allowance commencing in 2020 as follows:

- Two hundred dollars (\$200.00) per year for full time employees actively at work who
 are required to wear a uniform, to be paid in two installments of one hundred dollars
 (\$100.00) in May and November each year;
- Ten cents per hour (\$0.10/hour) for part time employees who are required to wear a uniform paid biweekly

The parties agree that the payment of a uniform allowance for full time employees shall be prorated based on completed months of service. For clarity, employees who are on an *Employment Standards Act* leave are considered to be actively at work.

BETWEEN:

THE CORPORATION OF THE CITY OF LONDON (DEARNESS SERVICES)

- and -

UNIFOR, LOCAL 302

Re: Summer Students - PSW/HCA

WHEREAS the Corporation may offer temporary full-time hours for up to 15 Personal Support Worker/Health Care Aide (PSW/HCA) positions in accordance with the Letter of Understanding re: Temporary Full Time Hours;

AND WHEREAS the Union has agreed that the Corporation may seek individuals outside the bargaining unit to temporarily work as PSW/HCAs from May 1 to mid-September;

NOW THEREFORE the parties agree as follows:

- The Corporation may seek to employ individuals outside of the bargaining unit to work either full time temporary hours (in accordance with paragraph 3) or part time casual temporary hours in the PSW/HCA classification up to a maximum of 15 positions.
- The parties agree that individuals employed in accordance with paragraph 1 above shall be hired as casual part time employees but for a fixed term starting no earlier than May 1 and ending no later than mid-September.
- The parties agree that individuals who are hired in accordance with paragraph 2 above may be assigned to temporary full time hours for PSW/HCA as outlined in the LOU re: Temporary Full time Hours.
- 4. The parties agree that individuals hired under the terms of this Letter of Understanding will be considered as casual part time employees under the Collective Agreement. With the exception that Article 21 and Article 12.9 will not apply to individuals hired in accordance with paragraph 2.
- 5. The Corporation shall meet with the Union the first week of March of each year to discuss the number of Full-time and Part-time PSW/HCAs to be off on vacation for the applicable vacation period. After the discussion, the Corporation will determine the number of Full-time and Part-time PSW/HCAs to be off on vacation for the applicable vacation period and advise the Union.